

An Industry Perspective About Quality Management System (QMS) for Drug-Device Combination Products



Author: EFPIA/MQEG/GMP Working Group on DDC • Date: 04/11/2019 • Version: Final

A Pharma Industry perspective on the relationship between European Medical Device Regulations MDR 2017/745 and the Pharmaceuticals Quality System, as set forth in Eudralex Volume 4 Chapter I and in other Pharmaceutical Quality System documents such as ICH Q10 and 21 CFR (US-FDA), aimed at facilitating the implementation of MDR 2017/745 for Pharma Industry manufacturing and/or marketing Drug Device Combination Products in Europe.



Foreword

This Document was developed by the EFPIA-MQEG/GMP Working Group on Drug-Device Combinations (DDC).

Authors:

Guy Godeau,^{1,*} Hervé Varenne^{2,*}, Karin Herzog³, Audrey Clarembeaux⁴, Mike Barnett⁵, Susan Neadle⁶, Amanda Matthews⁷, Ruth Murtagh^{8,} Joe Nagle⁹, Mark Sakitis¹⁰, Torsten Kneuss¹¹

* Corresponding author email: Guy.godeau@ucb.com

* Corresponding author email: Varenne_herve@lilly.com

About EFPIA

The Manufacturing & Quality Expert Group (MQEG) is a specialised group within the European Federation of Pharmaceutical Industries and Associations (EFPIA), which is recognised as the leading (bio)pharmaceutical association in Europe. Within MQEG, a Good Manufacturing Practices (GMP) Working Group (WG) addresses quality and compliance aspects related to Drug-Device Combination (DDC) products.

The EFPIA initiative for GMP aspects of DDCs was driven by a group composed mainly with Quality Experts in Development and Quality of DDCs, and supported by 3 Regulatory Experts, representing the majority of EFPIA company members; The composition of the WG is provided on next page.

"Pharma Industry" or "Pharma Company" mentioned in the title and throughout this paper refers to EFPIA member companies.

This document was published under the authority of the EFPIA-MQEG Committee on 31 October 2019. It represents an industry association perspective and does not confer any legal aspect, nor any immunity to its user (Person or Legal Entity). The perspective is built on the study of the regulation, industry discussion and consensus, and is not set in stone or agreed by the Regulators (EMA, CAMD) at this time.



EFPIA Initiative for GMP Aspects of DDCs – Composition of the Working Group

<u>Chairman</u>	<u>Company</u>	Title / Function
Guy Godeau ¹	UCB	Director, Corporate Quality Assurance GMP Compliance & Auditing
Hervé Varenne ²	Lilly	Global Quality Advisor, Medical Devices & Combination Products
Working Group Members	Company	Title / Function
Karin Herzog ³	Boehringer Ingelheim	Senior Quality Manager and Global Management System Owner for Medical Devices and Drug Device Combination Products
Audrey Clarembeaux ⁴	UCB	Quality Lead Development Medical Devices
Mike Barnett ⁵	Astra Zeneca	Global Associate Director - Device Quality
Susan Neadle ⁶	J&J	Global R&D Quality Lead
Amanda Matthews ⁷	Pfizer	Senior Director, Global Regulatory Affairs, Global CMC – Devices
Ruth Murtagh ⁸	GSK	Director, Global Product Quality Office
Joe Nagle ⁹	Amgen	Director Quality Assurance, Devices
Mark Sakitis ¹⁰	MSD	Director, Quality Systems and Compliance
Torsten Kneuss ¹¹	Bayer	Pharma Quality Medical Devices
Alexander Valenca	Sanofi	Head of Compliance, Device Development Quality Operations
Mike Wallenstein	Novartis	Executive Director QA, Senior Compliance Officer
Michael Karl Ledinegg	Sandoz	Senior Manager Global Development QA External Collaboration
Claude-Alain Piguet	Merck Serono	Head of Device Quality Management
Sylvain Hallynck	UCB	Head of Medical Device QA
Serkan Oray	UCB	Head of Medical Device



Executive Summary

The present industry perspective represents the current level of understanding of QMS requirements for MDR 2017/745 which would be pertinent to Pharma Industry designing, developing, manufacturing and marketing Drug-Device Combination (DDC) products, taking into account the clarifications brought by EMA in its draft Guidance on Quality Requirements for DDC Products (dGQR-DDC), EMA/CHMP/QWP/BWP/259165/2019 dated May 29, 2019.

This document aims to clarify and facilitate the implementation of MDR requirements for QMS by Pharma Industry being considered the legal manufacturer of the DDC, and not being considered as the legal manufacturer of the Device ⁽¹⁾. Drug Device Combination products (DDCs) *intended to administer a medicinal product* are in the scope of this document. The DDCs, for which the Device has the primary mode of action, is out of scope of this document.

To achieve this objective, the document provides a Pharma Industry perspective on one holistic Quality System approach addressing the requirements provided in the pertinent quality system in Regulation (EU) 2017/745 on Medical Devices (MDR) and the pharmaceutical quality system (PQS) requirements set forth in Eudralex Volume 4, Chapter I, Pharmaceutical Quality System (PQS), and in other relevant quality system requirements such as ICH Q10, US-FDA 21 CFR part 4, 21 CFR Part 820 and ISO 13485 (1). We conclude that the intention of all these expectations on Quality Systems have similar intentions in the specific scope they are written for and can be combined in one QMS in a company addressing all expectations for DDCs.

The Working Group members performed a mapping of MDR 2017/745 towards a better understanding of the quality system requirements for Pharma Industry manufacturing and/or marketing DDCs. The outcome of this mapping was further discussed and consolidated in workshops held in October 2018 and February 2019. Where appropriate, clarification and alignment obtained on specific MDR Article with other Industry Associations are mentioned in this document. Indeed, in February 2019, EFPIA initiated a collaboration on a set of 16 questions with MedTech Pharma Platform (MPP) and Medicines for Europe (MFE). Feedbacks from both MPP & MFE were obtained and resulted in alignment and clarification obtained for 8 questions, need for clarification from Stakeholders (Regulators) for 7 questions, and 6 points for further advocacy, at least for EFPIA.

The target audience of this document are the EFPIA members. There are essentially two reasons for that:

1) At this stage the regulatory pathway and how to submit CMC data, for DDC intended to administer a medicinal product, are not fully clear; This document identifies 15 so-called 'Requests for clarification from Stakeholders', and 10 'EFPIA points for advocacy', as described in Section 5 (Appendix 1) and 6. These are or will be conveyed to Stakeholders using different media as underlined in Section 6. Therefore, clarification of GMP and more specifically QMS requirements remain open to several options depending on the responses to EFPIA requests.

2) The EFPIA Quality Expert Working Group for GMP Aspects of DDCs (EFPIA WG) makes recommendations for QMS. However, these do not represent the only possible approach, and the EFPIA WG would not like to interfere in the current EMA consultation process on its dGQR-DDC. Therefore, the recommendations should remain internal to Industry.

This reflection paper is structured with 6 Sections:

- Section 1 describes the scope of the document,
- Section 2 provides a description of the challenges and considerations for the Quality Management System (QMS) when implementing the MDR 2017/745 for Drug-Device Combinations (DDC),



- Section 3 explains the precautions recommended by the EFPIA WG,
- Section 4 indicates regulatory document references for terminology and definition,
- Section 5 introduces the way Appendix 1 was built, which provides the applicability and relevance of MDR requirements as a function of the type of combination product, the link between MDR 2017/745 and PQS (Eudralex Vol 4) and the key messages to interpret MDR 2017/745 within the context of DDCs,
- Section 6 provides an exhaustive list of the 'Request for clarification from Stakeholders (Regulators)' and the 'EFPIA points for advocacy'.
 These questions and points for advocacy are collected by EFPIA-EBE and will be provided to EMA either through the EMA consultation process on its draft Guidance on Quality Requirements for DDC, or through EFPIA-EBE reflection papers.

Actually, it is expected that a Pharma Company manufacturing and distributing DDCs (but that is not the legal manufacturer of the Device) would not be audited by a Notified Body (NB), but instead inspected by the relevant Supervisory Authority) for medicinal products. Therefore, a PQS would be sufficient. However, there are certain quality requirements in MDR to consider:

1. Despite the fact that DDCs are registered as medicinal products, there are Articles and Annexes in MDR 2017/745 to comply with when commercialising DDCs, especially Annex I (GSPR),

2. There is a need to confirm the applicability / non-applicability of MDR 2017/745 beyond CE marking for Non-Integral DDCs mainly,

3. There is a need for the Pharma Company to check for appropriate understanding and interfacing between PQS and Medical Device QMS aspects,

4. Understanding key MDR requirements for QMS will facilitate the Pharma Company to design, develop, manufacture and distribute DDCs,

5. Understanding MDR requirements makes it easier for the Pharma Company to ensure appropriate quality oversight on Device legal Manufacturers or their Authorized Representatives if located outside of EU, and to establish quality assurance agreement (QAA) that fits for purpose.

(1) For legal manufacturer of Medical Devices, the Authors refer the readers to the excellent document establishing the relationship between MDR 2017/745 and ISO 13485:2016 as described in CEN/TR 17223:2018: Guidance on the relationship between EN ISO 13485: 2016 (Medical devices – Quality management systems – Requirements for regulatory purposes) and European Medical Devices Regulation and In Vitro Diagnostic Medical Devices Regulation.



Table of Contents

1. Scope
2. Challenges and considerations for Quality Management System (QMS) when implementing the MDR 2017/745 for Drug-Device Combinations (DDC)
2.1 Definition of Drug-Device Combination Product
2.2 QMS for Drug-Device Combination Product (Device used as a drug delivery system)10
3. Limitations
4. Terms and Definitions1
5. A Pharma Industry perspective on the relationship between European Medical Device Regulations MDR 2017/745 and the Pharmaceutical Quality System (Introduction to Appendix 1)
6. Summary of the Requests for Clarifications from Stakeholders (Regulators) and EFPIA Points for Advocacy

Appendix 1	.31
------------	-----



1. Scope

This reflection paper provides a Pharma Industry perspective on the relationship between the pertinent quality system requirements in Regulation (EU) 2017/745 on Medical Devices (MDR) and the pharmaceutical quality system requirements set forth in Eudralex Volume 4, Chapter I, Pharmaceutical Quality System (PQS), and in other relevant quality system documents such as ICH Q10 and US-FDA 21 CFR part 4.

It aims to clarify and facilitate the implementation of MDR requirements for QMS by Pharma Industry and raises a series of points for clarification and advocacy to promote harmonisation of QMS requirements for designing, developing, manufacturing and marketing Drug-Device Combination (DDC) products in Europe. DDCs *intended to administer a medicinal product* are in the scope of this document. Combination products, for which the Device has the primary mode of action, are not within scope for this document.

This industry perspective represents our current understanding of which QMS requirements in MDR 2017/745 that are pertinent to Pharma Industry for DDC products. Account has been taken of the clarifications made by the EMA in its draft Guidance on Quality Requirements for DDC Products (dGQR-DDC), EMA/CHMP/QWP/BWP/259165/2019 dated May 29, 2019.

Our perspective on the relationship between quality system requirements of both domains (Medical Devices and Medicinal Products) is provided in Appendix 1 in six (6) different sections, as suggested by the MDR Flowchart developed by MedTech Europe⁽²⁾:

-	MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	Appendix 1 – Pages 32 to 66
-	MDR Process Part II - Clinical Evaluation	Appendix 1 – Pages 67 to 73
-	MDR Process Part III – Device Classification and Conformity Assessment	Appendix 1 – Pages 74 to 85
-	MDR Process Part IV - Registration Process	Appendix 1 – Pages 86 to 89
-	MDR Process part V - Post-Marketing Surveillance	Appendix 1 – Pages 90 to 92
-	MDR Process Part VI - Obligations of the Other Economic Actors	Appendix 1 – Pages 93 to 97

(2) MedTech Europe - Overview of requirements under the Medical Devices Regulation 2017/745/EU on Medical Devices – Flowchart – December 2017

(3) It is recognized that the meeting between EFPIA/EBE and TEAM-NB WG on July 5th 2019 also provides some insight with regards to NB expectations for SI DDC products and clinical data expectations as part of the NB assessment to meet Article 117.



2. Challenges and considerations for Quality Management System (QMS) when implementing the MDR 2017/745 for Drug-Device Combinations (DDC)

2.1 Definition of Drug-Device Combination Product (DDC)

Neither the MDR 2017/745 nor the Medicinal Product Directive 2001/83/EC on medicinal products for human use define explicitly a Drug Delivery Device Combination (DDC) to the extent that the reader would know clearly which requirement of the MDR would or would not apply to the Device when combined with a Medicinal Product. There is a need for clarification in order to assess the impact on the QMS.

The recent EMA draft Guidance on Quality Requirements for Drug-Device Combination (EMA dGQR-DDC) provides more clarity. However,

- There are remaining questions and potential issues to address for both (Single-) Integral and Non-Integral DDCs, which could have an impact on the Pharma Industry QMS
- EMA dGQR-DDC document primary focus are is for DDCs falling within the definition of Article 1(9) of the MDR (Line 99 of EMA dGQR-DDC), which leaves CE Marked Devices for Non-Integral DDC with questions related to the applicability of MDR beyond CE marking,
- The draft clarifications provided in dGQR-DDC require confirmation through finalisation of the EMA guidance document.

a) Definition of Single-Integral and Non-Integral DDC

MDR 2017/745 states that products combining a medical device and a medicinal product follow specific rules (Article 1 (8) and Article 1(9)). These specific rules provide subject matter and scope of the Regulation for Devices when used in combination with a Drug and introduce the concept of Integral and Single Integral Drug-Device Combination Products.

Although the concept of 'Single Integral' is well defined in Article 1 (9), the terminology 'Integral' described in Article 1 (8) is not defined in the regulations and leaves the 'Non-Integral' Drug-Device Combination Product undefined as well.

The following aspects related to **DDC definition** need clarification or confirmation:

- What does 'Integral' mean and when does a DDC product become Integral? When placed on the market, or at the point of use? Article 1 (8) seems to leave the possibility that the concept of integral product could be considered at the point of use, i.e., 'when put into service'. However, it seems that this is only applicable to combination products for which the Device has the principal mode of action. The second part of Article 1 (8) states



that, if the Drug has the principal mode of action, then the combination product is considered as a medicinal product under Directive 2001/83/EC or Regulation (EC) No 726/2004. And in that case, it seems necessary to refer to Article 1 (9), second sub-part, that defines and limits the concept of **single-Integral at the point when the product is placed on the market.** The EMA guidance document (Lines 67-69 EMA dGQR-DDC) appears to confirm this interpretation.

What is a Non-Integral DDC? The concept of Non-Integral combination product is not defined in MDR 2017/745. However, EMA dGQR-DDC provides a clear definition of Non-Integral DDC in the Introduction (Lines 81-83) by stating that Devices not falling within the scope of Article 1(8) and 1(9) of the MDR should be CE marked (Lines 86-87). As per MDR 2017/745, Article 1 (9) first subpart, Any device which is intended to administer a medicinal product as defined in point 2 of Article 1 of Directive 2001/83/EC shall be governed by this Regulation, without prejudice to the provisions of that Directive and of Regulation (EC) No 726/2004 with regard to the medicinal product. The last sentence requires clarification in order to identify which Articles of MDR 2017/745 would or would not apply to Pharma Industry manufacturing (combining) Non-Integral DDC beyond CE marking. This question is pertinent mainly for the MDR Articles defining requirements for Labelling, Distribution, UDI for Traceability and Post-marketing surveillance.

b) QMS impact arising from the potential applicability of MDR 2017/745 to the Device when combined with the Drug

Subsequent to the issues with the definition of DDC (With Drug having the principal mode of action), the *applicability of MDR to the Device WHEN COMBINED with the drug* was questioned.

The EMA dGQR-DDC brings clarification in order to avoid a potential conflict of requirements between MDR 2017/745 and Directive 2001/83/EC or Regulation (EC) No 726/2004. Indeed, *The core precept* of the EMA guideline is that the Competent Authority for the regulation of medicines (CA) will evaluate the <u>device-specific aspects</u> of safety and performance relevant to the quality, safety and efficacy of the medicinal product, and that, as applicable, the NB will assess the relevant General Safety and Performance Requirements (GSPRs) (Lines 155-157):

- For (Single) Integral (4) Product, if the Device is not CE marked and if it would require the involvement of an NB, then a Notified Body opinion (NBOp) on the conformity of the device with the relevant GSPRs should be issued by an appropriately-designated NB (Lines 151-153);
- For Non-Integral DDCs: The Device must be CE marked in accordance with the MDR. Where a CE marked device for the administration of the
 medicinal product is co-packaged or is referred to in the SmPC of a marketing authorisation, additional information may need to be provided by the
 applicant (To CA) with regards to the device if the device may have an impact on the quality, safety and/or efficacy of the medicinal product (Lines 157160).

The issuance of EMA dGQR-DDC brings much clarification, but also raises new questions, mainly related to "How to provide the information to CA" and to the respective roles of NBs vs CAs. These questions are raised in Section 2.2 below "QMS for Drug Delivery Devices combined with Medicinal Product", herebelow, and in the core part of the document, i.e., Section 5 "A Pharma Industry perspective on the relationship between European Medical Device Regulations MDR 2017/745 and the Pharmaceutical Quality System".

(4) Note: EMA uses two terminologies Integral, which includes Single-Integral, and Non-Integral (Lines 61-69). The scope of this document includes Single-Integral DDC and Non-Integral Co-packaged DDC



For ease of review, Section 6 provides an exhaustive list of the 'Request for clarification from Stakeholders (Regulators)' and the 'EFPIA points for advocacy'.

2.2 QMS for Drug-Device Combination Product (Device used as a drug delivery system)

In Europe, the publishing of MDR 2017/745 implies new QMS considerations for Pharma Industry designing, manufacturing and/or marketing DDC.

a) Application of Standards & ISO 13485:2016 for QMS

While it is clear that Notified Bodies (NBs) will assess the conformity of the Device alone against requirements set forth in MDR 2017/745, the question was raised whether Pharma Industry would require EN ISO 13485:2016⁽⁵⁾ certification for the manufacture of DDCs. It is also expected that CA GMP inspectors, not Notified Bodies (NBs) will inspect the development and manufacture of Single-Integral DDCs and the co-packaging of drugs with CE marked devices as per cGMP requirements.

Adoption and certification of ISO 13485:2016 may be considered as an asset for the Pharma Company, but is not an absolute regulatory requirement, at least not in EU.

Moreover, EMA has stated clearly in its dGQR-DDC under section 4.1 "Application of standards (Lines 169-171), that *Compliance of a DDC with relevant Ph. Eur. chapter(s) or monograph(s) should be demonstrated. Ph.Eur. requirements and European and ICH guidance take precedence over ISO standards.*

If a Pharma Company has a CE marked Device that is part of a DDC and if this Pharma Company is the legal manufacturer of the Device, then there are additional benefits of having ISO 13485 to comply with the MDR QMS requirements; The relationship between EN ISO 13485:2016 ⁽⁵⁾ and MDR 2017/745 and IVDR 2017/746 is provided in a Guidance document issued under the authority of Technical Committee CEN/TR 17223:2018 entrusted to quality management and corresponding general aspects for medical devices.

(5) EN ISO 13485:2016, Medical devices - Quality management systems - Requirements for regulatory purposes.



b) MDR challenges for Pharma Industry QMS (PQS)

The new challenges brought by MDR 2017/745 depend on the type of DDC product and experience of the Pharma Company with Medical Devices. The following challenges could impact Pharma Industry QMS and/or quality/technical workload:

a. For both Single-Integral and Non-Integral DDCs:

- i. A reduced number of Notified Bodies (NBs), since MDR 2017/745 brings more stringent requirements on their designation. QMS impact for Pharma Company: Securing new a NB requires extensive vendor management activity.
 - For Non-Integral DDC, each Pharma Company should check with the legal manufacturer of the Device whether NBs, ensuring the CE certification of Device under MDD, re-apply for their role of NB under MDR 2017/745 requirements or not, and for which Device designation. The need for NB is key to ensure that the Device is followed for Post-Market Surveillance (PMS) and for Post-Market Clinical aspects (PMCF). The need for re-certification of Device by May26, 2020 is only required for new Device or for Device with modified intended purpose, major design or production changes classification, and after May 26, 2020 as a function of CE certificate expiration date. Vendor management activity for new Third Party (NB) requires substantial technical and quality works.
 - 2. For Single-Integral DDC, Pharma Company should also contract a NB for any new DDC for which a NBOp would be required to support file submission after May 26, 2020. This would also be the case for changes to existing devices on the market that have a change and would require a variation to the file after 26 May 2020. See also Paragraph b.i. here-below.

ii. The MDR reclassifies certain devices and has a wider scope (MDR Annex VIII).

QMS impact for Pharma Company: Change of Device classification imply technical and quality works associated with re-certification activities.

- 1. For Non-Integral DDC, each Pharma Company should check whether the CE marked Device class is modified. If the Device risk classification is modified, then the Pharma Company should liaise with the Device manufacturer to determine any potential impact pertinent (PMS-PMCF) to the revision of the Quality Agreement approved with the Device manufacturer.
- 2. EMA has also mentioned the use of MDR 2017/745 classification rules for Single-Integral DDC in its Q & A document EMA/37991/2019 issued on February 28, 2019, and in its dGQR-DDC (Lines 411-412). However, the practical aspects associated with device classification for NBOp is unclear at this stage.
- iii. The MDR introduces a clinical evaluation consultation procedure for Class IIb active devices, intended to administer and/or remove a medicinal product, as referred to in Section 6.4 of Annex VIII (Rule 12), by an independent expert panel (Article 54). QMS impact for Pharma Company: Not known at this stage, since clarification about the applicability of this requirement is requested from EMA



- 1. For both Single-Integral and Non-Integral DDC, each Pharma Company should for new CE marked Devices (used as parts of DDCs) check if they may fall in Class IIb as per MDR rule 12 and *in cases of doubt as to the proposed classification of the device according to the MDR, it is recommended that an opinion be sought from a medical device CA* (Lines 162-163 of EMA dGQR-DDC).
- 2. The use of classification rules for a device, mentioned by EMA, does not mean that a DDC classified IIb would need to be reviewed by the Expert Panel (clinical evaluation consultation) since DDCs are registered as medicinal products.

b. For Single-Integral DDCs:

If the device part would not be CE marked, and would require the involvement of an NB, then the MDR 2017/745 introduces a Notified Body Opinion (NBOp) made obligatory for the conformity of the device part of DDC product with the relevant general safety and performance requirements (GSPR) set out in MDR 2017/745 Annex I issued by a NB designated in accordance with that Regulation for the type of device in question (MDR 2017/745 Article 117).
 QMS impact for Pharma Company: Need to comply with the relevant GSPR of MDR Annex I.

c. For Non-Integral DDCs:

The MDR will also provide increased transparency, with information on devices and studies being made public.
 QMS impact for Pharma Company: Not known at this stage. Clarification about the applicability of requirements of MDR 2017/745 beyond CE marking is requested from stakeholders (EMA, DG HEALTH, DG GROW).

The new European Database for Medical Devices – EUDAMED – will play a central role in making data available and increasing both the quantity and quality of data (Article 33): This Article is pertinent for legal Manufacturers of CE marked Devices; The Authors of this document consider that Pharma Industry manufacturing Non-Integral DDC would not be impacted by these requirements. However,

- This would require confirmation by both EMA & the competent authority for Medical Device, i.e., at EC DG GROW level.

- Accessing EUDAMED is free for public information. Would Pharma Company need access to additional information to assess the use of a Device? In such a case, how would Pharma Company require a Registration Number (SRN) to be registered as an economic actor under MDR?

ii. A new **Unique Device Identification (UDI) system** (Article 27) will significantly enhance the traceability and the effectiveness of postmarket safety-related activities, and **Information** to supply with the device (Article 23).

QMS impact for Pharma Company: Not completely understood at this stage. Clarification about the applicability of requirements of MDR 2017/745 beyond CE marking is requested from stakeholders (EMA, EC Health, EC Growth).

This requirement is not pertinent to Single-Integral DDC and the extent of its applicability to Non-Integral DDC is not clear yet. This would require clarification from both EMA & the National Competent Authority for medical devices, i.e., at EC DG GROW level.



iii. MDR introduces specific requirements for **Importers of Medical Devices** (Article 13, 16, 30 & 31): This is applicable to Pharma Industry directly responsible for the importation of Devices into EU. EFPIA, MPP & MFE are aligned on the applicability of this MDR requirement.

QMS impact: Need to comply with registration of economic operators (Articles 30 & 31) and with requirements for Importers (Article13). Obligations of manufacturers of Medical Device should not apply to importers if some specific conditions are met (See Article 16, below).

For importation of Non-Integral DDC into EU, the applicability of MDR beyond CE marking remains a request for clarification from stakeholders, similar to points under c.i. & c.ii.

iv. MDR introduces specific requirements for Distributors of Medical Devices (Article 14);

QMS impact: The requirements for Distributor would mean, among other requirements, that Pharma Company marketing DDC, wholesaler, hospital & pharmacy would need to check for UDI and other aspects as per Article 14 (2). The potential issue is developed further here-below.

Potential issue with Article 14:

a) *Regulatory-wise*: As soon as a device is put into the folding box together with the drug, the distributor role according to the MDR should not apply anymore. In that moment the individual medical devices with CE mark received becomes part of the "Drug" from a physical and market authorization perspective.

b) Quality-wise:

- First of all it needs to be said that the distributor in the MDR is understood to be a legal-entity based view, not something which applies to a group of companies as a whole. That said there are typically multiple legal entities within a pharmaceutical company which constitute a separate distributor and would have to fulfill the requirements of Article 14. In addition there are 3rd party companies which are mandated to further distribute medical devices on behalf of the pharmaceutical company.

- When it comes to medical devices co-packaged with the drug (e.g. needles, transfer devices, alcohol swabs, dry powder inhaler etc.), it is neither appropriate nor possible that every legal entity in the distribution chain fully performs all obligations of Article 14, e.g. to check/confirm the compliance is not possible anymore. This would require to open individual drug packages, which in many cases have to be sealed with a tamper evident seal in Europe.

- Instead EFPIA would consider the medical devices not falling under the distributor obligations of MDR Art 14 anymore, as soon as the folding box with the device and the drug has been manufactured. Pharma company would follow cGMP & cGDP regulatory requirements.



- That means that the pharmaceutical manufacturing site (1st distributor) which receives the device from the manufacturer would perform all the necessary checks and would maintain files for documentation. Furthermore compliance to medicinal product directives for Good Distribution Practices (2013/C 343/ 01) and to Regulation 2016/161 for serialization and safety measures by each distribution actor would ensure patient safety and protection along the supply chain. Affiliate(s), wholesaler, hospital and pharmacy would not repeat those checks, however they would of course forward complaints to the legal manufacturer of the device (via the MAH and the 1st distributor).

- One argument why this process could be sufficient is that the relevant distributors in that chain are either affiliates of the pharmaceutical company which fall under the global quality management system or they are contractors which are covered by supplier qualification process.

Result of the survey performed with other industry associations:

EFPIA contacted other Industry Associations (MedTech Pharma Platform & Medicines for Europe). There is no clear alignment among the Industry Associations. EFPIA requires clarification from the Stakeholders since Non-Integral DDC is registered as a medicinal product. See Section 6 for "Clarification request" and "EFPIA point for advocacy". EFPIA advocates for non-applicability of this Article to Pharma Industry distributing DDCs, since it would duplicate the requirements set forth in Directive for Good Distribution Practices (2013/C 343/ 01) and the COMMISSION DELEGATED REGULATION (EU) 2016/161 for traceability (Serialisation code) and safety measures for medicinal products. This would require clarification from both EMA & the competent authority for Medical Device, i.e., at EC DG Growth level.

v. MDR introduces potential requirements for **Distributors and Importers to comply with manufacturer obligations** (Article 16); As described in the rationale (37) of the MDR, the Article 16 has been introduced to define conditions which allow relabelling and repackaging as required to support parallel trading.

QMS impact: Applicability of manufacturer obligations to the Pharma Company would represent a huge impact, since the whole MDR would then apply. There are two different scenarios, developed here-below, in order to prevent applicability of manufacturer obligations to the Pharma Company manufacturing Co-packaged DDC.

Scenario 1: The Pharmaceutical Company receives devices in bulk packaging from the manufacturer. In this case the further distribution of the device(s) in that packaging is not intended, but the bulk packaged devices are intended to be separated by the Pharmaceutical Company. Bulk packaging also means that the boxes are typically unlabeled, or labelled with the minimum requirements as per MDR 2017/745. This should be carried out in such conditions that the original condition of the device cannot be affected by this operation (Article 16(2)). Complying with the other paragraphs of MDR Article 16 (1) would prevent the Pharma Company from Medical Device manufacturer obligations. **QMS impact** would therefore be limited to appropriate Quality Agreement to delineate responsibilities between manufacturer and pharma company.



Scenario 2: The pharmaceutical company receives devices in a large sales unit (e.g. cannulas, or single-use syringes) which is already on the market. This folding box of the sales unit has separate labelling (incl. information not covered by the individual unit labelling) and includes one single IFU. Now, the Pharmaceutical Company separates this sales unit by co-packaging only one device out of this sales unit.

- This "changes of pack size" would mean that the Pharmaceutical Company as distributor would become the legal manufacturer of the Device. The only reason for this separation of sales unit is that the marketing authorization for the drug requires a "change of pack size", for the intended procedure with the drug, and not to allow for marketing of the device itself in certain countries.

- The exemption clauses of Art. 16(2) might not apply and would lead the Pharma Company to endorse the manufacturer obligations.

- In order to prevent to fall into the requirements of Article 16, the Pharma Company should perform the separation under (Purchasing) agreement with the legal manufacturer.

- QMS impact could be minimized if the legal manufacturer provides the medical device in a pack size which does not require separation, with either an IFU included in the pack or an IFU master in the required languages which can be included in the PIL without any modification.

Considering the various MDR challenges, the Authors recommend Pharma Companies, who currently work according to the existing Medical Device Directive (MDD) or are using their Pharmaceutical Quality Systems (PQSs) with no specific Medical Device QMS, to check for adequate coverage of pertinent MDR requirements by its current PQS/QMS. There is no one QMS solution fit for all. It is up to each Pharma Company to adapt its QMS in order to meet the pertinent MDR requirements for both Single-Integral and Non-Integral DDCs.

The following endpoints should be kept in mind when revising the Pharma Company PQS:

- For Single-Integral DDC, the QMS helps to comply with General Safety and Performance Requirements (GSPRs) set forth in MDR Annex I,
- For Non-Integral DDC, the intended use set forth in the Conformity Assessment for the CE marked Device is maintained, and if not, a new conformity assessment is performed as per MDR towards NB certification.
- The overall combination of Device and Medicinal Product is taken into consideration, at the point to place the DDC on the market and at the point of use.



The Authors consider that the following Device specific QMS areas should receive particular attention from the Pharma Company when revising its PQS:

- Product realisation (Design Controls): This process, as described in ISO 13485:2016, or in MDR Annex IX, should be used right at the beginning of the development of a new Single Integral Product and/or Non-Integral DDC, and should include Usability Engineering to address the design requirements at point of use of DDC.
- Purchasing Controls,
- Management Responsibilities (especially Management Representative),
- Effective and pro-active CAPA system,
- Product-specific risk management, ISO14971 and/or ICH Q9. The role of the risk management and especially benefit-risk analysis becomes more and more important, with input from the post-market surveillance system.

This document explores the options in sufficient detail and provides insight how to assess the potential impact of MDR on QMS. The decision to select QMS options is always a choice not a regulatory requirement.

This document provides an industry perspective about:

The applicability and relevance of MDR requirements as a function of the type of drug-device combination product (Single Integral or Non-Integral DDC)
 The possibility to use the Pharmaceutical Quality System. It outlines its potential limitations and suggests opportunities to adapt it at company level in order to adequately address the requirements set forth in MDR 2017/745.

3. Limitations

Not all Articles of MDR 2017/745 or it's annexes were reviewed and analysed by the Authors. The selection of Articles and Annexes was based on the best technical and quality judgment of the Authors after mapping of MDR 2017/745 towards a better understanding of the quality system requirements for Pharma Industry manufacturing and/or marketing DDCs.

This is a living document which will evolve with the issuance of interpretative guidance documents by the Regulators and experience and feedback from EFPIA Pharma Industry members. Any question, suggestion or feedback will be welcomed by the Authors.

Last but not least, the recommendations provided in Table 1 represent the consensus within the Working Group after mapping of the MDR text and discussion. However, these recommendations are not the sole possible approach for interpreting the MDR requirements or for designing a QMS for DDCs. It is up to each Pharma Company to design an adequate PQS that meets regulatory requirements.



4. Terms and Definitions

For the purposes of this document, the terms and definitions used in Eudralex Volume 4, in MDR 2017/745 and in EMA draft Guidance on Quality Requirements for DDC apply.

5. A Pharma Industry perspective on the relationship between European Medical Device Regulations MDR 2017/745 and the Pharmaceutical Quality System (Introduction to Appendix 1)

Appendix 1 provides the Pharma Industry perspective on key requirements of the European Regulations on Medical Devices (MDR EU 2017/745), together with commentary on the extent to:

- Which PQS could be used in alignment with Eudralex Volume 4, or with ICH Guidelines
- How to interpret MDR clause (Key message)
- Which clarification is needed from the Regulators (Request for clarification from Stakeholders)
- Advocate for a position on the applicability of a requirement (Point of advocacy)
- Recommend to Pharma Industry-specific MDR interpretation for DDCs
- When pertinent, how the requirement is interpreted in US-FDA 21CFR part 4, mainly because:
 - o It would underline similarities or divergences in requirements,
 - o 21 CFR Part 4 suggests for a streamlined QMS approach, combining PQS and Medical Device QMS, when considering DDC products.

The perspective on the relationship between quality system requirements of both domains (Medical Devices and Medicinal Products) is provided in six (6) different sections, as suggested by the MDR Flowchart developed by MedTech Europe⁽⁶⁾:

-	MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	Appendix 1 – Pages 32 to 66
-	MDR Process Part II - Clinical Evaluation	Appendix 1 – Pages 67 to 73
-	MDR Process Part III – Device Classification and Conformity Assessment	Appendix 1 – Pages 74 to 85
-	MDR Process Part IV - Registration Process	Appendix 1 – Pages 86 to 89
-	MDR Process part V - Post-Marketing Surveillance	Appendix 1 – Pages 90 to 92
-	MDR Process Part VI - Obligations of the Other Economic Actors	Appendix 1 – Pages 93 to 97

(6) MedTech Europe - Overview of requirements under the Medical Devices Regulation 2017/745/EU on Medical Devices – Flowchart – December 2017



6. Summary of the Requests for Clarifications from Stakeholders (Regulators) and EFPIA Points for Advocacy

6.1. Requests for Clarifications from Stakeholders (Regulators)

Legend: SI = Single-Integral / NI = Non-Integral

MDR Process	1400	-	Requests for Clarifications from Stakeholders (Regulators)	Decision about
	MDR	Туре	Requests for Clarifications from Stakeholders (Regulators)	
Part I -	Article	of		process to
Obligations of	or	DDC		convey the
the	Annex	(SI or		requests to
manufacturer		NIor		Stakeholders
and Annex I for				
General Safety		SI &		
and		NI)		
Performance				
requirements				
	Article 1 (9)	NI	The part of the first sentence of Article 1 Section 9 (", without prejudice to the provisions of that Directive (2001/83/EC) and of Regulation (EC) n° 726/2004 with regard to the medicinal product.") lacks clarification about which Articles of the MDR 2017/745 would not apply to the Medical Device when co-packaged with the medicinal product.	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs
	Article 8	SI & NI	Article 8 might presume that conformity with harmonized standards related to system or process is a must. EFPIA will work with Stakeholders to ensure a correct interpretation of this article, which should be: Device products and / or systems being in conformity with relevant harmonized standards as published in the Official Journal of the European Union would facilitate conformity assessment, but Ph.Eur. requirements and European and ICH guidance-take precedence over ISO and other international standards.	Consultation process on EMA dGQR-DDC – Comment to lines 170-171 (Chapter 4. General considerations for integral DDC)
	Article 10 (3)	NI	It is not clear whether the Pharma company would need to contribute to Post Market Clinical Follow-up (PMCF) as per Annex XIV. Complying to MDR 2017/745 beyond CE marking should be clarified by EMA for Co-packaged DDC.	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs



MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
	Article 10 (8)	SI & NI	 MDR specifies archiving timeframes of 10 years (15 years for implantable) after the last device put on the market. This would be a confirmed requirement for the device legal manufacturer only. For Pharma company manufacturing a DDC , this would be defined by the registration type, i.e., by medicinal product directives if registered as a medicinal product. EFPIA will work to clarify MDR 2017/745 requirements beyond CE marking. 	New set of questions for EMA Q&A format document
	Annex I Chapter II Section 10.4.3.	SI & NI	Will Stakeholders look for harmonization with international standards ICH or with EMA existing guidelines on Phtalates (EMA guideline EMA/CHMP/SWP/362974/2012)	Consultation process on EMA dGQR-DDC – Comment to lines 170-171 (Chapter 4. General considerations for integral DDC) – See Article 8 here-above
	Annex I Chapter III (Sections 23.1., 23.2, 23.3 and 23.4)	SI	EFPIA will work with EMA to define the labelling requirements specific to the device part, especially for Single-Integral DDC.	EFPIA-EREG Reflection paper for labelling of SI DDC



MDR Process Part II - Clinical Evaluation ⁽⁷⁾	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
	Article 61 (1)	SI	 Annex I does not mention Clinical Evaluation nor Article 61. Clarification is required as Clinical Evaluation does not necessarily mean clinical data, and DDC clinical data do not necessarily include the use of the single integral device. What level of evidence would be required by NB, as a function of the device component risk class and on the available clinical evaluation ? 	Consultation process on EMA dGQR-DDC – Comment to lines 249-253 and 254-259 (Chapter 5.2.Module3.2.P. Drug Product) + Meeting between EFPIA/EBE and TEAM-NB WG on July 5 th 2019
	Article 61 (1)	NI	Beyond CE marking, clarification is required with regards to requirement for Clinical Evaluation, and MDR 2017/745 Article 61 especially, as DDC clinical data do not necessarily include the use of the same CE marked device.	Consultation process on EMA dGQR-DDC – Comment to lines 611 (Chapter 7 – Bridging to devices used in



		clinical
		development)

(7) It is recognized that the meeting between EFPIA/EBE and TEAM-NB WG on July 5th 2019 also provides some insight with regards to NB expectations for SI DDC products and clinical data expectations as part of the NB assessment to meet Article 117.

MDR Process Part III – Device Classification and Conformity Assessment	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
	Article 86 (3)	NI	However, it is not clear yet to what extent Pharma Company would need to provide data Periodic Safety Update Reports (PSUR) to legal manufacturer for the Devices used in Non-Integral DDCs	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs
	Annex IX Chapter I,Section 4.10	SI & NI	Change of an approved device: EFPIA would like clarifications with regards to the translation of DDC change types in EMA variation procedures (Type IA, IB and II).	Consultation process on EMA dGQR-DDC – Comment to lines 615-640 (Chapter 8 Lifecycle management) +
				EFPIA Reflection Paper on substantial changes (Initiated)
	Article 117	SI	For the submission of a new Single Integral DDC product after 26 May 2020, would a relevant certificate issued by a notified body allowing the manufacturer to affix a CE marking to the medical device under the MDD be acceptable regulatory-wise providing that the certificate is still valid ?	EMA dGQR-DDC – Comment to lines 145-147 (Chapter 4. General



MDR Process Part III – Device Classification and Conformity Assessment	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
				considerations for integral DDC)

MDR Process	MDR Article or Annex	Type of	Requests for Clarifications from Stakeholders (Regulators)	
Part IV – Registration				
Process			Annex	or NI or
		SI & NI)		
	NA	NA	NA	



MDR Process Part V – Post- Market Surveillance (PMS)	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
	Article 83	SI & NI	While the reporting pathway would be clear for DDCs registered as medicinal products, a Post-Market Surveillance System is required as part of Annex I (GENERAL SAFETY AND PERFORMANCE REQUIREMENTS), Chapter I General Requirements, Section 3.e. Risk-Management system. This could be interpreted as indirect need to establish a PMS system according Article 83 also for DDC products as described in MDR 2017/745 article I(9) and article 117. EFPIA will work with the Stakeholder to obtain guidelines with regards to applicability of MDR Article 83 to SI & NI DDCs.	Consultation process on EMA dGQR- DDC – Comment to lines 641-642 (Chapter 8 Lifecycle management) + For NI DDC: EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs



MDR Process Part VI – Obligations of the Other Economic Actors (Distributors and Importers)	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	Requests for Clarifications from Stakeholders (Regulators)	Decision about process to convey the requests to Stakeholders
	Article 14	NI	Applicability of MDR 2017/745 Articles, such as Article 14, beyond CE marking would require clarification from Regulators. It is EFPIA opinion that there is no need for Pharma Company distributing Non-Integral Co-Packaged DDC to be considered as Distributor of Device under MDR Article 14, since the CE marked Device is co-packaged with the medicinal product and must be distributed as per GDPs, including as per Regulation 2016/161 for Safety Features. Therefore Complying with requirements set forth in Article 14 is ensured by default. This is valid for all Sections of Article 14 (1, 3, 4, 5 & 6), with a specific point for advocacy related to UDI check (See Article 14 Section 2 in Section 6.2. EFPIA Points for Advocacy).	EFPIA-MPP-MFE Reflection Paper for Non-Integral DDCs
	Article 13	NI	EFPIA will work with EMA and Medical Device Regulators to clarify the applicability of MDR 2017/745 Article 13 for Pharma Company importing Non-Integral, co-packaged, DDC.	EFPIA-MPP-MFE Reflection Paper for Non-Integral DDCs
	Article 16	NI	EFPIA will work with EMA and Medical Device Regulators to clarify the applicability of MDR 2017/745 Article 16 for Pharma Company importing Non-Integral, co-packaged, DDC.	EFPIA-MPP-MFE Reflection Paper for Non-Integral DDCs



6.2. EFPIA Points for Advocacy

MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	EFPIA Points for Advocacy	Decision about process to convey the requests to Stakeholders
	Article 10 (3)	SI	MDR 2017/745 Annex I (GSPR) does not refer to Article 61 (Clinical Evaluation). Therefore Single Integral DDC are excluded from this MDR requirement	The meeting between EFPIA/EBE and TEAM-NB WG on July 5 th 2019 also provides some insight with regards to NB expectations for SI DDC products and clinical data expectations as part of the NB assessment to meet Article 117.
	Article 10 (9)	SI & NI	The description of the QMS requirements does not include expectations with regards to device change management. Pharma Industry would advocate for major change to the device being covered in the QMS using recognized guidance document, which would be aligned with conditions described in paragraph 4.9 of Annex VII of the MDR, i;e., changes to the approved type of Device, to its design, to its intended purpose or claim made for it, or to any substance incorporated in or used for the manufacture of the Device	- For SI DDC: Question 15 of EFPIA-EBE set of questions communicated to EMA on June 12, 2019



MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	EFPIA Points for Advocacy	Decision about process to convey the requests to Stakeholders
				+ Consultation process on EMA dGQR-DDC – Comment to lines 615-640 (Chapter 8 Lifecycle management)
	Article 10 (9)	SI	Article 83 is not included in Annex I, and is therefore not applicable to Single Integral DDC. The post-market surveillance system should comply with medicinal product directives	EFPIA-EBE Reflection Paper for Substantial Changes of SI DDC + Consultation process on EMA dGQR-DDC – Comment to lines 641-642 (Chapter 8 Lifecycle management)
	Article 10 (9)	NI	Article 83 is applicable to the Device legal manufacturer. Pharma Company shall report adverse event and complaint caused by the device component to the National Competent Authority of the medicinal product, and would need to communicate this type of information to the Device manufacturer	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs



MDR Process Part II - Clinical Evaluation ⁽⁷⁾	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	EFPIA Points for Advocacy	Decision about process to convey the requests to Stakeholders
	Article 61 (1)	SI & NI	It is EFPIA advocacy point that there is no need to apply MDR 2017/745 Article 61 requirement for DDC product that are supported by drug clinical data delivered with the device of the DDC product.	- For SI DDC: See MDR Process part I - For SI & NI DDC: Consultation process on EMA dGQR-DDC – Comment to lines 611 (Chapter 7 – Bridging to devices used in clinical development)

(7) It is recognized that the meeting between EFPIA/EBE and TEAM-NB WG on July 5th 2019 also provides some insight with regards to NB expectations for SI DDC products and clinical data expectations as part of the NB assessment to meet Article 117.



MDR Process	MDR	Туре	EFPIA Points for Advocacy	Decision about
Part III – Device	Article	of		process to
Classification	or Annex	DDC		convey the
and Conformity		(SI or		requests to Stakeholders
Assessment		NIor		Stakenoiders
		SI &		
		NI)		
	Article 56 (1 & 2)	SI & NI	EFPIA advocate for a simplified procedure, when considering no or few changes, also on a guidance or rule, using a risk-based approach for NB to decide on the minimum validity of a certificate. This would promote harmonization as much as possible.	EFPIA Reflection Paper on substantial changes (Initiated)
	Article 117	SI & NI	EFPIA would like clear process about how fees and timelines are made public, how fees are derived, and how Device legal Manufacturer and Pharma Industry can use these information in the NB engagement process (Cfr Art. 53, engagement of 1 NB at a time)	To be defined at EFPIA-MQEG level - Meeting with NB and/or MDCG (? TBC)



MDR Process Part IV – Registration Process	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	EFPIA Points for Advocacy	Decision about process to convey the requests to Stakeholders
	Article 29 (1)	NI	The PQS ensures that the device information is traceable; EFPIA advocates for UDI being affixed on the device itself or its primary packaging, not on the secondary packaging of the DDC, similar to US-FDA 21 CFR Part820. The UDI will be checked and documented in DDC batch records and SmPC. Non-Integral DDCs registered as medicinal product must indeed comply with serialisation for safety measure and traceability along the distribution chain.	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs
	Article 30 (2)	NI	On top of EFPIA request for streamline approach of GDP for Non-Integral DDC, i.e., with limited application of Article 29, EFPIA advocates for non- applicability of Articles 14 & 30 to Pharma Company distributing Non-Integral DDCs.	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs

MDR Process	MDR	Type of	EFPIA Points for Advocacy		
Part V – Post-Market	Article or Annex	DDC (SI			
Surveillance (PMS)	Annex	Annex	Annex	or NI or	
		SI & NI)			
	NA	NA	NA		



MDR Process	MDR Article or Annex	Type of DDC (SI or NI or SI & NI)	EFPIA Points for Advocacy	Decision about process to convey the requests to Stakeholders
Part VI – Obligations of the Other Economic Actors (Distributors and Importers)	Article 14	NI	Requirements set forth under (a), (b), (c) & (d) are normal requirements for a Pharma Company operating under the PQS: Purchasing a CE marked Device, combining it with a medicinal product and placing it on the market require controls at reception of the Device, during manufacturing of the Non-Integral DDC and before QP batch certification. Therefore EFPIA would recommend an interpretative guidance, which would allow for DDC product, registered as medicinal product, to be exempt of the requirements set forth in Article 14. If Pharma Company would be considered as Distributor under MDR Article 14, then EFPIA would advocate for UDI check performed at the DDC assembly site <u>only</u> , and not along the supply chain (Wholesalers, hospital,), since it would not be feasible (UDI on primary packaging), or would require UDI on each packaging (First, second, third,).	EFPIA-MPP- MFE Reflection Paper for Non- Integral DDCs



APPENDIX

A Pharma Industry perspective on the relationship between European Medical Device Regulations MDR 2017/745 and the Pharmaceutical Quality System

Table 1 provides the Pharma Industry perspective on key requirements of the European Regulations on Medical Devices (MDR EU 2017/745), together with commentary on the extent to:

- Which PQS could be used in alignment with Eudralex Volume 4, or with ICH Guidelines
- How to interpret MDR clause (Key message)
- Which clarification is needed from the Regulators (Request for clarification from Stakeholders)
- Advocate for a position on the applicability of a requirement (Point of advocacy)
- Recommend to Pharma Industry specific MDR interpretation for DDCs
- When pertinent, how the requirement is interpreted in US-FDA 21CFR part 4, mainly because:
 - o It would underline similarities or divergences in requirements,
 - o 21 CFR Part 4 suggests for a streamlined QMS approach, combining PQS and Medical Device QMS, when considering DDC products.

The perspective on the relationship between quality system requirements of both domains (Medical Devices and Medicinal Products) is provided in six (6) different sections, as suggested by the MDR Flowchart developed by MedTech Europe⁽¹⁾:

-	MDR Process Part I - Obligations of the manufacturer and Annex I for General Safety and Performance requirements	Table 1 – Pages 32 to 36
-	MDR Process Part II - Clinical Evaluation	Table 1 – Pages 37 to 43
-	MDR Process Part III – Device Classification and Conformity Assessment	Table 1 – Pages 44 to 54
-	MDR Process Part IV - Registration Process	Table 1 – Pages 55 to 58
-	MDR Process part V - Post-Marketing Surveillance	Table 1 – Pages 59 to 61
-	MDR Process Part VI - Obligations of the Other Economic Actors	Table 1 – Pages 62 to 66

(1) MedTech Europe - Overview of requirements under the Medical Devices Regulation 2017/745/EU on Medical Devices – Flowchart – December 2017 QMS guidance for Pharma company manufacturing and/or marketing DDC



MDR Article reference	MDR text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufacturer a	nd General Safet	y and Performance	Requirements	
Article 1 Section 9	Any device which is intended to administer a medicinal product as defined in point 2 of Article 1 of Directive 2001/83/EC shall be	Annex I General Safety	EU Directive 2001/83/EC	Links with PQS 1) EU MDR defines 2 categories of drug the medicinal product has the primary n	g-device combination products for which node of actions:
Subject matter and scope	governed by this Regulation, without prejudice to the provisions of that Directive and of Regulation (EC) No 726/2004 with regard to the medicinal product. However, if the device intended to administer a medicinal product and the medicinal product are placed on the market in such a way that they form a single integral product which is intended exclusively for use in the given combination and which is not reusable, that single integral product shall be governed by Directive 2001/83/EC or Regulation (EC) No 726/2004, as applicable. In that case, the relevant general safety and performance requirements set out in Annex I to this Regulation shall apply as far as the safety and performance of the device part of the single integral product are concerned.	& Performance Requirements (GSPR)	21 CFR 3.2(e), defines combination product. 21 CFR 4 refers to 21 CFR 3.2(e)	 Single Integral drug delivery device Annex I of MDR is required , and other drug delivery devices, for marked. For the latest, EMA dGQR-DDC introdu which includes Co-packaged DDC and although this terminology used in the US DDC Line 85, 86: "or where the Product 	es, for which only compliance with or which MDR applies and must be CE aces the terminology "Non Integral", so called 'Cross labelled DDC", S is not used by EMA (EMA dGQR- t Information (SmPC and Package used with the medicinal product but the inal product: bstances which may be used in or th a view to restoring, correcting or erting a pharmacological, o making a medical diagnosis. as "administration to patients, route of roper administration, method of ders:



MDR Article reference	MDR text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufacturer a	nd General Safet	ty and Performance	Requirements	
				of the Medical Device Regulation (MDR when co-packaged with the medicinal p	t) would not apply to the Medical Device roduct.
				the applicability of the regulation/guidar and 21CFR820 in its essential points.	ation of combination products regarding nce. 21CFR4 is a mix of 21CFR210/211
Article 2 Section 30 Definition of Manufacturer	'Manufacturer' means a natural or legal person who manufactures or fully refurbishes a device or has a device designed, manufactured or fully refurbished, and markets that device under its name or trademark	NA	COMMISSION DIRECTIVE 2003/94/EC of 8 October 2003 Article 2 Definitions 3. US FDA : 21CFR820 o)	Links with PQS EU GMP: 'Manufacturer' means any person engaged in activities for which the authorization referred to in Article 40(1) and (3) of Directive 2001/83/EC or the authorization referred to in Article 13(1) of Directive 2001/20/EC is required; Article 40 1. Member States shall take all appropriate measures to ensure that the manufacture of the medicinal products within their territory is subject to the holding of an authorization. This manufacturing authorization shall be required. Key messages: 1) The manufacturer of the drug delivery device part of a Single Integral DDC might not fall into the definition of MDR, nor of EU Directive 2003/94/EC. The manufacturer of the device component might comply with ISO 15378 "Primary packaging	Links with PQS The manufacturer of the CE marked device component of co-packaged DDC (= "Non-integral" DDC) falls under the definition of MDR EU745/2017. The manufacturer of the co-packaged DDC falls under the definition of EU Directive 2003/94/EC, unless the Pharma company performs activities that fall under the scope of MDR Article 16 (1a), (2b), (3) & (4). See development of MDR Article 16. As per MDR Article 14, the Pharma company manufacturing and marketing a co-packaged DDC would fall under the requirements of Distributor (See development of MDR Article 14). Key messages: Two causalities come together to define a manufacturer as per MDR:



MDR Article reference	MDR text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)					
MDR PROCESS PART I – Obligations of the Manufacturer and General Safety and Performance Requirements										
				materials for medicinal products — Particular requirements for the application of ISO 9001, with reference to good manufacturing practice (GMP)". 2) The manufacturer of the DDC falls under the definition of EU Directive 2003/94/EC.	The manufacturer always markets the device under its name or tradename: However if the legal entity only markets the device without having any design, manufacturing or refurbishment steps the legal entity is not defined as manufacturer but potentially as distributor (Article 14). Point of advocacy EFPIA would recommend an interpretative guidance, which would allow for DDC product, registered as medicinal product, to be exempt of the requirements set forth in Article 14, or limited the requirements to added value aspect, like the check of UDI at the manufacturing site only. See MDR Process part VI- Obligations of the Other Economic Actors					



MDR Article reference	MDR text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)				
MDR PROCESS PART I – Obligations of the Manufacturer and General Safety and Performance Requirements									
				US FDA 21cfr820.1 o):					
				Manufacturer means any person who designs, manufactures, fabricates, assembles, or processes a finished device. Manufacturer includes but is not limited to those who perform the functions of contract sterilization, installation, relabelling, remanufacturing, repacking, or specification development, and initial distributors of foreign entities performing these functions.					
				According to FDA: manufacturer does n name or trademark.	ot market the Medical Device under its				



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)					
MDR PROCESS PART I – Obligations of the Manufacturer and General Safety and Performance Requirements										
Article 5, sections 2 & 3 Placing on the market and putting into service	A device shall meet the general safety and performance requirements set out in Annex I which apply to it, taking into account its intended purpose. Demonstration of conformity with the general safety and performance requirements shall include a clinical evaluation in accordance with Article 61.	Annex I General Safety and Performance Requirements (GSPR)	EU Directive 2001/83/EC	Key message: Single-Integral DDC is registered as medicinal product which must comply with MDR Annex I (GSPR). The latest does not refer to Article 61.	Key messages: The device part of the Co- packaged DDC must be used within the intended purpose. If the Pharma company used the device for a purpose which was not intended by the device manufacturer, then the Pharma company becomes manufacturer as per MDR definition and needs to comply with manufacturer obligations as per MDR Article 16. See development of MDR Article 16 (MDR PROCESS PART VI – OBLIGATIONS OF THE OTHER ECONOMIC ACTORS)					
Article 8 Use of harmonized standards & the monographs of the European Pharmacopoeia	Devices that are in conformity with the relevant harmonized standards, or the relevant parts of those standards, the references of which have been published in the Official Journal of the European Union, shall be presumed to be in conformity with the requirements of this Regulation covered by those standards or parts thereof.	NA	Eudralex Vol 4	Request for clarification from StakeholdersArticle 8 might presume that conformity with harmonized standardsrelated to system or process is a must.EFPIA will work with Stakeholders to ensure a correct interpretationof this article, which should be: Device products and / or systemsbeing in conformity with relevant harmonized standards aspublished in the Official Journal of the European Union wouldfacilitate conformity assessment, but Ph.Eur. requirements andEuropean and ICH guidance-take precedence over ISO and otherinternational standards.						



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirer	nents	
	The first subparagraph shall also apply to system or process requirements to be fulfilled in accordance with this Regulation by economic operators or sponsors, including those relating to quality management systems, risk management, post-market surveillance systems, clinical investigations, clinical evaluation or post-market clinical follow-up ('PMCF'). References in this Regulation to harmonized standards shall be understood as meaning harmonized standards the references of which have been published in the Official Journal of the European Union.			 2. Article 8, section 2, considers the m European Pharmacopoeia as harmoniz those related to interaction between me materials used in devices containing su <u>Recommendations to Pharma Industry</u> a) As outlined in the introduction, Pharmand/or marketing DDC products could n System (PQS) and complement the QN requirements of device quality system. not deemed a requirement. Applying pr facilitate the compliance with some of the b) Note: Some Rest of the World (ROW Taiwan, require an ISO 13485 certificat December 5th, 2018 a transition from the ISO 13485:2016. ISO 13485:2016 has a globally in the world. 	ted standards, especially edicinal products and uch medicinal products. The company manufacturing rely on the Pharma Quality AS with key specific Certification to ISO 13485 is rinciples of ISO 13485 does he MDR requirements. () countries, , like Canada, ion. FDA announced also on the 21 CFR part 820 to
Article 10 Section 2 General Obligation of the Manufacturer Risk Management System	Manufacturers shall establish, document, implement and maintain a system for risk management as described in Section 3 of Annex I.	Annex I General Safety and Performance Requirements (GSPR)	Eudralex 4 – Chapter 1 PQS – QRM 1.12 & 1.13 ICH Q9 ISO EN 13458 ISO 14971 EMA dGQR-DDC	Link with PQS EU GMP provides the expectations wit the quality of product and link with the refer to ICH Q9 for the framework (How See Review for Annex I Section 3 Risk As per EMA dGQR-DDC lines 241, 242 integral DDC, while the EMA documen integral DDC section (Lines 506-507) t of product development information i	protection of patient, and v to do). Management System. RM is expected for single it mentions risk in non- io indicate that the amount



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the	Manufacturer and Genera	I Safety and Performance Require	ements	
				reflect the risk of the device to impac quality of the medicinal product.	t the safety, efficacy &
				 <u>Recommendation to Pharma Industry</u> The development of the DDC needs to be embarked in risk management process, which should comply with ISO 14971 at ICH Q9. Risk management is part of the overall review of the product with the device. See also notes about ISO 14971 and EN 62366 here-below. ISO 14971, Medical Devices – Application of risk management medical devices. If the results of risk analysis indicate that use errors could cause serious harm to the patient or the device us then the manufacturer should apply appropriate human factors usability engineering processes. 	
				ISO EN 62366 "Human factors": Shou relation with ISO 14971.	ld be a consideration , in
				US FDA 21 CFR 820.30: As part of their Design conduct a risk analysis that includes th device use and the measures implement	e risks associated with



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requiren	nents	
Article 10 Section 3 General Obligation of	Manufacturers shall conduct a clinical evaluation in accordance with the requirements set out in Article 61 and Annex XIV, including a PMCF.	Annex XIV Clinical Evaluation and Post-Marketing Follow-up (CE & PMCF)	EU Directive 2001/83/EC USFDA 21CFR820.30 Design controls	Point of advocacy MDR Annex I (GSPR) does not refer to Article 61. Therefore Single Integral DDC are excluded from this MDR requirement.	Key messages: No additional requirements if Pharma company use the CE marked device within its intended use.
the Manufacturer Clinical evaluation including Post Marketing Clinical Follow- up (CE & PMCF)				 Recommendation to Pharma Industry a) Pharma company need to take into consideration Annex I, Chapter II (10.3). if the devices are intended to administer medicinal products they shall be designed and manufactured in such a way as to be compatible with the medicinal products concerned in accordance with the provisions and restrictions governing those medicinal products and that the performance of both the medicinal products and of the devices is maintained in accordance with their respective indications and intended use. b) Clinical aspects of the device component are to be embarked in Design control for DDC product, as outlined in the Introduction. c) It is assumed that drug clinical data cover the device constituent for safety and performance, and therefore Art 61 would not apply. Let us pay attention 	Request for clarificationfrom StakeholderIt is not clear whether thePharma company wouldneed to contribute toPMCF as per Annex XIV.Complying to MDRbeyond CE markingshould be clarified byEMA for Co-packagedDDC.Recommendation toPharma IndustryClinical aspects of thedevice component are tobe embarked in Designcontrol for DDC product,as outlined in theIntroduction.US FDA21 CFR 820.30: As part oftheir Design Controls,



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General Sa	afety and Performance Requirem	ients	
				 no other requirements than Annex I would be necessary for Single Integral products, as Risk management for Benefit/Risk ratio is required. d) For well-established device, Clinical Evaluation Report (CER) should contain Drug Clinical Data and Scientific Data & Literature on the device in order to support clinical claims. e) For novel device, clinical investigation and data would be required. US FDA 21 CFR 820.30: As part of their Design Controls, manufacturers conduct a risk analysis that includes the risks associated with device use and the measures implemented to reduce those risks. Clinical evaluation aspects should be embarked in Design Control of the DDC product. 	risk analysis that includes the risks associated with device use and the measures implemented to reduce those risks. Clinical evaluation aspects should be embarked in Design Control of the DDC product.



Article 10 Section 8 General Obligation of the Manufacturer Technical documentation	Manufacturers shall keep the technical documentation, the EU declaration of conformity and, if applicable, a copy of any relevant certificate, including any amendments and supplements, issued in accordance with Article 56, available for the competent authorities for a period of at least 10 years after the last device covered by the EU declaration of conformity has been placed on the market. In the case of implantable devices, the period shall be at least 15 years after the last device has been placed on the market. Upon request by a competent authority, the manufacturer shall, as indicated therein, provide that technical documentation in its entirety or a summary thereof. A manufacturer with a registered place of business outside the Union shall, in order to allow its authorised representative to fulfil the tasks mentioned in Article 11(3), ensure that the authorised representative has the necessary documentation permanently available.	NA	Eudralex Vol 4 Chapter 4 documentation 4.11 Batch documentation 4.12 Other type of documentation ICH M4 : The Common Technical Document	 Link with PQS In contrast to the registration procedures for Medicinal products with ICH CTD format, the registration procedure of MDs product completely differs and is not comparable. Pharma company should comply with Eudralex Vol 4 and ICH M4 requirements, as MDR is not applicable to manufacturer of Single Integral DDC beyond Annex I GSPR, nor to Co-packaged DDC assuming that Pharma would not be the device manufacturer, nor would fall under Article 16 requirements (Cases when general obligations of the manufacturer apply to other economical actors). a) Eudralex Vol 4:Chapter 4 documentation 4.11 Specific requirements apply to batch documentation which must be kept for one year after expiry of the batch to which it relates or at least five years after certification of the batch by the Qualified Person, whichever is the longer. For investigational medicinal products, the batch documentation must be kept for at least five years after the completion or formal discontinuation of the last clinical trial in which the batch was used. 4.12 For other types of documentation, the retention period will depend on the business activity which the documentation supports. b) ICH M4 : The Common Technical Document:
--	---	----	---	--



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General Sa	afety and Performance Requirem	nents	
Article 10 Section 9 General Obligation of the Manufacturer Quality Management System	Manufacturers shall ensure that procedures are in place to keep series production in conformity with the requirements of this regulation. Changes in device design or characteristics and changes in the harmonised standards or CS by reference to which the conformity of a device is declared shall be adequately taken into account in a timely manner. Manufacturers of devices, other than investigational devices, shall establish, document, implement, maintain, keep up to date and continually improve a quality management system that shall ensure compliance with this Regulation in the most effective manner and in a manner that is proportionate to the risk class and the type of device. The quality management system shall cover all parts and elements of a manufacturer's organisation dealing with the quality of processes, procedures and devices. It shall govern the structure, responsibilities, procedures, processes and management resources required to implement the principles and actions necessary to achieve	Annex I General Safety and Performance Requirements (GSPR) Annex XIV Clinical Evaluation and Post-Marketing Follow-up (CE & PMCF)	Eudralex Vol 4 Chapter I, Pharmaceutical Quality System (PQS) ICH Q10 Pharmaceutical Quality System implementation US FDA: 21CFR211 for GMP Medicinal products 21CFR820 for QMS for Medical Devices.	 Link with PQS For Pharma company designing, manufal DDC, a holistic approach for its QMS is the conformity assessment requirements review of the possible requirements is p Single Integral and Co-packaged DDC. The Publication of titles and references under European Union harmonisation less tandards for specific Medical Devices for includes EN ISO 13485:2016 Medical demanagement systems — Requirements This is a commonly acknowledged stand designing and manufacturing medical designing and manufacture of medicinal production. That is proportionate to the risk class. Section 1.4 A Pharmaceutical Qualit the manufacture of medicinal production (xii) Arrangements are in place for the planned changes and their approvataking into account regulatory notifier required; Art 6 of Directives 2003/94/EC and manufacturers to establish and imp pharmaceutical Quality System is uninterests of consistency with ICH Q 	necessary. All depend on s to comply with MDR. A rovided here below for of harmonised standards gislation contains ISO/EN technical requirements. This evices — Quality for regulatory purposes. dard for Manufacturers evices. the door to a holistic eral aspects: y of the company's activities compare to "in a manner s" ty System appropriate for cts should ensure that: he prospective evaluation of l prior to implementation cation and approval where 91/412/EEC require lement an effective ystem. The term ised in this chapter in the



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Require	ments	
	compliance with the provisions of this Regulation. The quality management system shall address at least the following aspects:			 ICH Q10 – Pharmaceuticals Quality S 1.5.1 Achieve Product Realisation to est maintain a system that allows the deliver quality attributes appropriate to meet th care professionals, regulatory authoritie approved regulatory filings) and other in customers. 1.5.2 Establish and Maintain a State of effective monitoring and control system and product quality, thereby providing a suitability and capability of processes. Obe useful in identifying the monitoring at covered in the QMS requirement expectations with regards to device charlindustry would advocate for major charl covered in the QMS using recognized would be aligned with high level conditied 4.9 of Annex VII of the MDR, i.e., charl Device, to its design, to its intended purt to any substance incorporated in or use Device. Recommendation to Pharma Industry is the conformity assessment requirement review of the possible requirements is p Single Integral and Co-packaged DDC. Pharma company to check its PQS and aspects from MDR QMS as described in the PAR substance incorporated in the PAR substance incorporated	Atablish, implement and ery of products with the e needs of patients, health es (including compliance with internal and external Control To develop and use s for process performance assurance of continued Quality risk management can nd control systems. Ats does not include ange management. Pharma ge to the device being guidance document, which ons described in paragraph ges to the approved type of pose or claim made for it, or ad for the manufacture of the ange and/or marketing necessary. All depend on ts to comply with MDR. A provided here below for EFPIA recommends complement it with specific



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	S PART I – Obligations of the Manufa	cturer and General S	afety and Performance Require	ments	
	The quality management system shall address at least the following aspects: (a) a strategy for regulatory compliance, including compliance with conformity assessment procedures and procedures for management of modifications to the devices covered by the system;	NA	ICH Q10 Sections 1.5.1. & 1.5.2. US FDA 21CFR820.20	from ISO 13485:2016, such as Design (expectations with regards to GSPR). US FDA 21 CFR Part 4 suggests for an integrate DDC products. Eudralex Vol 4 and ICH Q10 incorporate which require adaptation of the quality s activities, the risk to patient and needs f pharmaceutical manufacturers complyin 21CFR4 allows to implement an integra implementation of distinct aspects of 21 Link with PQS: A strategy for regulatory compliance sho requirements in PQS address these exp ICH Q10: 1.5.1 Achieve Product Realisation To est maintain a system that allows the deliver quality attributes appropriate to meet the care professionals, regulatory authoritie approved regulatory filings) and other in customers. 1.5.2 Establish and Maintain a State of effective monitoring and control systems and product quality, thereby providing a suitability and capability of processes. G be useful in identifying the monitoring and	ed QMS when considering e flexible QMS references system to the complexity of for compliance: For ng with 21CFR210/211 US ted QMS with the CFR 820. ould be established. Similar bectations: stablish, implement and ery of products with the e needs of patients, health s (including compliance with iternal and external Control To develop and use s for process performance usurance of continued Quality risk management can



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Ma	nufacturer and Genera	I Safety and Performance Require	ements	
				With regards to modifications to the de 4.9 provides some conditions to asses Authors have also noted the following Devices: NBOG_BPG_2014_3 (Subst However, nor MDR, nor EMAdGQR-D classifying changes to Device and/or to Request for clarification from Stake EFPIA would like clarifications with reg DDC change types in EMA variation po II), and their expectations on content a	s changes to a Device. The source document for Medical antial Change Guidance). DC provide clear criteria for o DDCs. holders pards to the translation of rocedures (Type IA, IB and
				Recommendations to Pharma Indus	try
				1. The Legal Manufacturer should ma assessment aligned with conditions Annex VII of the MDR.	
				2. Review of the change assessment the application process for engaging a undertaking Notified Body Opinions.	
				3. A change assessment should be ma change to device part of a DDC produc changes which do not fall under NB sc	t and not include PQS QMS
				US-FDA 21CFR820.20 Management Management responsibility requires ar regulatory compliance is considered in operations in order to obtain the marke	effective QMS while trinsic to the medical device



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	S PART I – Obligations of the Manufa				
	The quality management system shall address at least the following aspects: (b) identification of applicable general safety and performance requirements and exploration of options to address those requirements;	This refers to GSPR without explicit reference to Annex I	US FDA 21CFR820.30 ISO 13485	Recommendation to Pharma Industry The Authors recommend that applicable identified through implementation of De described in ISO 13485:2016 and 21CF	e requirements of Annex I be sign Control in QMS , as
	The quality management system shall address at least the following aspects: (c) responsibility of the management;	NA	Eudralex Vol. 4 Section 1.4 (v) Managerial responsibilities 21CFR820 ISO 13485	 Link with PQS Management responsibility is part of every QMS. Eudralex Vol 4 addresses this requirement, as described he below. EudraLex Vol 4 - Chapter I Pharmaceutical Quality System 1.4 A Pharmaceutical Quality System appropriate for the manufacture of medicinal products should ensure that: 1.5 Senior management has the ultimate responsibility to er effective Pharmaceutical Quality System is in place, adequa resourced and that roles, responsibilities, and authorities are defined, communicated and implemented throughout the organisation. Senior management's leadership and active participation in the Pharmaceutical Quality System is essent leadership should ensure the support and commitment of st levels and sites within the organisation to the Pharmaceutical Quality System. 	
				Chapter 2 Personnel:2.4 Senior management has the ultimat effective quality management system is	



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	SPART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirem	nents	
				quality objectives, and, that roles, resp are defined, communicated and implen organisation. Senior management shou that describes the overall intentions an related to quality and should ensure co effectiveness of the quality manageme compliance through participation in ma	nented throughout the uld establish a quality policy d direction of the company ntinuing suitability and nt system and GMP
	The quality management system shall address at least the following aspects: (d) resource management, including selection and control of suppliers and sub-contractors;	NA	EudraLex Vol 4 Chapter I Pharmaceutical Quality System Chapter V Production Chapter VII Outsourced activities ICH Q10 Section 2 Management responsibilities 21CFR820.20 Management responsibilities (2) Resources	 Link with PQS a) Eudralex Vol 4 addresses this required below. EudraLex Vol 4 - Chapter I Pharmace 1.4 A Pharmaceutical Quality System a manufacture of medicinal products shot (vi) Arrangements are made for the matthe correct starting and packaging matter monitoring of suppliers and for verifying the approved supply chain; (vii) Processes are in place to assure the outsourced activities. Chapter 4 documentation Technical Agreement are agreed betward acceptors for outsourced activities. Chapter 5 production Suppliers 5.27 The selection, qualification, approsuppliers of starting materials, together acceptance, should be documented as 	rement, as described here eutical Quality System appropriate for the uld ensure that: nufacture, supply and use of erials, the selection and g that each delivery is from the management of een contract givers and



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as lega manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Manuf	acturer and Genera	I Safety and Performance Require	ements	
				quality system The supporting evide material approval should be maintaine	
				Chapter 7 outsourced activities	
				b) ICH Q10 Section 2 Management r 2.7 Management of Outsourced Act Materials These requirements are similar to Eud	vities and Purchased
				Recommendation to Pharma Indust The Authors recommend that applicab (9) (d) be managed as per ISO 13485: Purchasing Controls.	le requirements of Article 10
				US FDA 21CFR820.20 Management (2) Resources. Each manufacturer shares resources, including the assignment of management, performance of work, are including internal quality audits, to meet the requirements of this part.	Il provide adequate trained personnel, for
				21CFR820. Subpart E—Purchasing Evaluate, select, maintain records clearly includes consultants	Controls
	The quality management system shall address at least the following aspects:	Annex I	ICH Q9	See Article 10 Section 2	



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirem	nents	
	(e) risk management as set out in Section 3 of Annex I;	General Safety and Performance Requirements (GSPR)			
	The quality management system shall address at least the following aspects: (f) clinical evaluation in accordance with Article 61 and Annex XIV, including PMCF;	Annex XIV Clinical Evaluation and Post-Marketing Follow-up (CE & PMCF)	EU Directive 2001/83/EC	See Article 10 Section 3	
	The quality management system shall address at least the following aspects: (g) product realisation, including planning, design, development, production and service provision;	NA	EudraLex Vol 4 Chapter I Pharmaceutical Quality System ICH Q10 Section 3.1 Lifecycle stage goals 3.1.1 Pharmaceutical Development	Link with PQS a) Eudralex Vol 4 Chapter I PQS 1.4 A Pharmaceutical Quality System a manufacture of medicinal products sho realisation is achieved by designing, p maintaining and continuously improvin consistent delivery of products with ap	buld ensure that: :i Product lanning, implementing, g a system that allows the
			US FDA 21CFR820: Subpart C—Design Controls	b) ICH Q10 Section 3.1 Lifecycle sta 3.1.1 Pharmaceutical Development <u>US-FDA</u> 21CFR820.70: Subpart G—I Controls	
	The quality management system shall address at least the following aspects: (h) verification of the UDI assignments made in accordance with Article 27(3) to all relevant devices and ensuring consistency	NA	NA	NA	Key messages: Potentially applicable to Co-packaged DDC. See EFPIA point for advocacy in MDR Process



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Manufa	cturer and General S	Safety and Performance Requirem	nents	
	and validity of information provided in accordance with Article 29;				Part IV – REGISTRATION PROCESS.
	The quality management system shall address at least the following aspects: (i) setting-up, implementation and maintenance of a post-market surveillance system, in accordance with Article 83;	NA	ICH Q10 3.2 PQS 3.2.1 Process Performance and Product Quality Monitoring System	Point of advocacy Article 83 is not included in Annex I, and is therefore not applicable to Single Integral DDC. The post-market surveillance system should comply with medicinal product directives.	Point of advocacy Article 83 is applicable to the Device legal manufacturer. Pharma Company shall report adverse event and complaint caused by the device component to the National Competent Authority of the medicinal product, and would need to communicate this type of information to the Device manufacturer.
	The quality management system shall address at least the following aspects: (j) handling communication with competent authorities, notified bodies, other economic operators, customers and/or other stakeholders;	NA	ICH Q10 3.2 PQS 3.2.1 Process Performance and Product Quality Monitoring System	Key messages: No modification of existing PQS	Key messages: Pharma company shall report adverse event and complaint caused by the device component to both the Device Manufacturer and the National Competent Authority of the medicinal product. Every communication to authorities for market action, registration etc. is



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Manufa	cturer and General S	Safety and Performance Require	ments	
					unique and need to be setup country by country.
	The quality management system shall address at least the following aspects: (k) processes for reporting of serious incidents and field safety corrective actions in the context of vigilance;	NA	NA	Key message: No modification of existing PQS	Key messages: Pharma company shall report adverse event and complaint caused by the device component to both the Device Manufacturer and the National Competent Authority of the medicinal product. General reporting strategies are already in place at every Pharma company. Recommendation to Pharma Industry Depending on the question who is the manufacturer additional reporting lines and timelines need to be
	The quality management system shall address at least the following aspects:	NA	ICH Q10 3.2 PQS	Link with PQS	established.



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	S PART I – Obligations of the Manufa	cturer and General S	Safety and Performance Require	ments	
	(I) management of corrective and preventive actions and verification of their effectiveness;		 3.2.2 Corrective Action and Preventive Action (CAPA) System 21CFR820 Subpart J - § 820.100 Corrective and preventive action. 	Based on ICH Q10 every pharmaceutic CAPA System in place. Recommendation to Pharma Industr EFPIA recommends to verify that the C includes proactive improvement CAPA from quality management reviews, and The idea is to reinforced the preventive ISO13485:2016)	Y APA system already , where pertinent decisions CAPA effectiveness check.
	The quality management system shall address at least the following aspects: (m) processes for monitoring and measurement of output, data analysis and product improvement.	NA	Eudralex Vol. 4 Section 1.4 US FDA 21CFR820.70 Production and process controls.	Link with PQS QMS for medicinal products is similar t Eudralex Vol 4 addresses this requirent below a) Eudralex Vol. 4 Section 1.4 (i) Product realisation is achieved by determined implementing, maintaining and continue that allows the consistent delivery of producity attributes; (ii) Product and process knowledge is relifecycle stages; US-FDA § 820.70 Production and production production processes to ensure its specifications.	nent, as described here- esigning, planning, ously improving a system oducts with appropriate managed throughout all Decess controls. evelop, conduct, control, and



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and Generation	al Safety and Performance Require	ements	
				(b) (2) Monitoring and control of process component and device characteristics of	
Article 10 Section 10 General Obligation of the Manufacturer Post Market Surveillance	Manufacturers of devices shall implement and keep up to date the post-market surveillance system in accordance with Article 83.	NA	EU Directives 2001/83/EC US-FDA 21 CFR822	Key messages:For Single Integral DDC, reporting to the competent authority would mean reporting to NCA / EMA only. There is no provision for double reporting as it is the case in Japan and US.If Articles 83_88 would not apply directly to Single Integral DDC products, the indirect relevance of these requirements remains, especially for high-risk class/type of device. MDR stipulates that the PMS system shall be proportionate to the risk class/type of the devices. A similar expectation is set forth by FDA. See recommendations here-below.See also MDR Process Part V – POST-MARKET SURVEILLANCE, VIGILANCE AND MARKET.Recommendation to Pharma Industry: PMS, as per MDR Articles 83-86, should be performed and keep	Key messages: MDR requirements apply to the Device Manufacturer only. Assuming that Pharma would not fall under the applicability of Article 16, this requirement does not apply directly to Pharma manufacturing a Co- packaged DDC. However, the indirect relevance of these requirements remains, especially for high-risk class/type of device. MDF stipulates that the PMS system shall be proportionate to the risk class/type of the devices. A similar expectation is set forth by FDA. See recommendations here- below. Adverse event (AE) caused by a Co-packaged



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	S PART I – Obligations of t	the Manufacturer and Genera	al Safety and Performance Require	ments	
				internally if not required by NB or by CAMD. Same recommendation for article 93 (PSUR data). EFPIA underlines that this requirement might depend on the device risk profile, as requires in the US by 21CFR822. See also note here-below pertinent to both Single Integral and Co-packaged DDC products.	DDC would be reported to the respective competent authority through 2 channels: The device part would be reported to CA for medicinal product by the MAH and to CAMD by the legal Device manufacturer of the device. These requirements should be described in the quality agreement approved between Pharma and medical device manufacturer. It would mean that Pharma need to report complaints to legal manufacturer of the device. See also MDR Process Part V – POST-MARKET SURVEILLANCE, VIGILANCE AND MARKET SURVEILLANCE.



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirem	nents	
Article 10 Section 12 General Obligation of the Manufacturer Ensure implementation of corrective actions	Manufacturers who consider or have reason to believe that a device which they have placed on the market or put into service is not in conformity with this Regulation shall immediately take the necessary corrective action to bring that device into conformity, to withdraw it or to recall it, as appropriate. They shall inform the distributors of the device in question and, where applicable, the authorised representative and	NA	Eudralex Vol. 4 Part 1 Chapter 8: Complaints, Quality Defects and Product Recalls US FDA 21 CFR Part 806: Corrections and Removals (Recalls)	Recommendation to Pharma Industry Pharma should continuously evaluated for understanding of the safety, quality the device. As per Medical Device QMS closes the loop with design control and/o Especially, ISO 14971 requires: The ma document and maintain a system to coll about the medical device or similar devit the post-production phase. E.g. To revie the risk management file is a reflection of US-FDA 21 CFR822 Post Market Surveillance according to 2 manufacturers to perform studies of hig have been granted 510(k) clearance or Link with PQS See Article 10 Section 10 here-above.	all device-related complaints and performance impact of requirements, this evaluation or manufacturing robustness. anufacturer shall establish, lect and review information ices in the production and ew risk/benefit and ensure of marketed device. 21CFR822 requires h-risk medical devices that
	importers accordingly. Where the device presents a serious risk, manufacturers shall				



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirem	nents	
Article 10 Section 13 General Obligation of	immediately inform the competent authorities of the Member States in which they made the device available and, where applicable, the notified body that issued a certificate for the device in accordance with Article 56, in particular, of the non-compliance and of any corrective action taken. Manufacturers shall have a system for recording and reporting of incidents and field safety corrective actions as described in Articles 87 and 88.	NA	Eudralex Vol. 4 Part 1 Chapter 8: Complaints, Quality Defects and Product Recalls US FDA 21 CFR Part 4 –	Link with PQS See Article 10 Section 10 here-above.	
Reporting incident and field safety corrective actions			Subpart B: Post-Marketing Safety Reporting for Combination Products		
Article 15 Section 1 Person Responsible for Pegulatory	Manufacturers shall have available within their organisation at least one person responsible for regulatory compliance who possesses the requisite expertise in the field of medical devices.	NA	EU Directive 2001/83/EC – Article 49 US FDA 21CFR820	Not directly applicable to Single Integral DDC manufacturer, since MDR Annex I is pertinent only.	Not applicable for Pharma manufacturing a Co- Packaged DDC assuming that requirements of Article 16 would not apply.
for Regulatory Compliance Requirements	The requisite expertise shall be demonstrated by either of the following qualifications: (a) a diploma, certificate or other			Link with PQS In Article 49 of Directive 2001/83 , the q the necessary experience of a QP is de	



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the Manufa	acturer and General	Safety and Performance Require	ements	
	evidence of formal qualification, awarded on completion of a university degree or of a course of study recognised as equivalent by the Member State concerned, in law, medicine, pharmacy, engineering or another relevant scientific discipline, and at least one year of professional experience in regulatory affairs or in quality management systems relating to medical devices; (b) four years of professional experience in regulatory affairs or in quality management systems relating to medical devices. Without prejudice to national provisions regarding professional qualifications, manufacturers of custom-made devices may demonstrate the requisite expertise referred to in the first subparagraph by having at least two years of professional experience within a relevant field of manufacturing.			 (2) "A qualified person shall be in poss certificate or other evidence of formal of completion of a university course of sture as equivalent by the Member State comperiod of at least four years of theoretic of the following scientific disciplines: phemedicine, chemistry, pharmaceutical of biology (). The course shall include the bearing upon at least the following bass. <u>Note:</u> Qualifications of PRRC and QP at experience type and duration in medicated demonstrated. <u>Recommendation to Pharma industration</u> as set forth in Article 15 Section pertinent to both type of DDC products. Therefore, EFPIA recommends to have Regulatory specialists to address the set in MDR, pertinent to drug administration device applications, in order to make the quality and regulatory Pharma quality set Pharmacovigilance. This recommendation is also justified for PDC products. 	 qualifications awarded on ady, or a course recognized acerned, extending over a cal and practical study in one harmacy, medicine, veterinary henistry and technology, heoretical and practical study ic subjects: (See Art 49) are comparable, but al devices should be Y: ectly to Pharma , PRRC a 3 (See here-below), are systems and Quality & pecific requirements set forth n devices and companion he appropriate link with other systems, including br Pharma marketing DDC



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General	Safety and Performance Require		
				US FDA 21CFR820 Management rep	resentative
Article 22 System and procedure packs Sections 1, 2 & 3 Basic requirements	 Natural or legal persons shall draw up a statement if they combine devices bearing a CE marking with the following other devices or products, in a manner that is compatible with the intended purpose of the devices or other products and within the limits of use specified by their manufacturers, in order to place them on the market as a system or procedure pack: (a) other devices bearing the CE marking; (b) in vitro diagnostic medical devices bearing the CE marking in conformity with Regulation (EU) 2017/746; (c) other products which are in conformity with legislation that applies to those products only where they are used within a medical procedure or their presence in the system or procedure pack is otherwise justified. 	NA	NA	Industry association position (With the Pharma Platform and Medicines for Eureceived and supported by EFPIA (Reserved and supported by EFPIA (Reserved and supported by EFPIA (Reserved and support of a	arope (MFE), opinions sponses to EFPIA Survey – rug Delivery Device tire DDC presentation is ective because the medicinal so provisions for procedure of the devices) would be et us make sure the intended tion product", "System" or of drugs or liquids. ded use does not include the system or procedure pack



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirer	nents	
Annex I – Gener	al Safety and Performance Require	ments			
Chapter I General requirements Section 3 Risk management system	Manufacturers shall establish, implement, document and maintain a risk management system. Risk management shall be understood as a continuous iterative process throughout the entire lifecycle of a device, requiring regular systematic updating. In carrying out risk management manufacturers shall: (a) establish and document a risk management plan for each device; (b) identify and analyse the known and foreseeable hazards associated with each device; (c) estimate and evaluate the risks associated with, and occurring during, the intended use and	NA (Annex I)	EU Eudralex Vol 4 Chapter I – PQS EU Medicinal Directive 2001/83/EC – Risk Management for Pharmacovigilance ICH Q9 & Q10 US-FDA	Link with PQS: For Medicinal product Risk Managemer Risk Management (ICHQ9) and Pharm Management (DIRECTIVE 2001/83/EC a) The general principles set forth is IC Risk Management are the same as for analyse, evaluate, report and mitigate. b) The principles for Risk Management (DIRECTIVE 2001/83/EC) are similar t safety & efficacy: Risk Management Sy pharmacovigilance activities and interv characterise, prevent or minimise risks product, including the assessment of th activities and interventions. Companies are required to submit a ris to the European Medicines Agency (El marketing authorisation. To help applic guidance on how to submit RMPs. It sl - A medicine's safety profile; - How its risks will be prevented or the	hacovigilance Risk C): CH Q10 & ICH Q9 for GMP Medical Devices: identify, t Pharmacovigilance to ICH Q9, but with a focus on ystem is a set of ventions designed to identify, a relating to a medicinal he effectiveness of those sk management plan (RMP) MA) when applying for a cants, EMA developed hould include information on:



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Require	ments	
	during reasonably foreseeable misuse; (d) eliminate or control the risks referred to in point (c) in accordance with the requirements of Section 4; (e) evaluate the impact of information from the production phase and, in particular, from the post-market surveillance system, on hazards and the frequency of occurrence thereof, on estimates of their associated risks, as well as on the overall risk, benefit-risk ratio and risk acceptability; and (f) based on the evaluation of the impact of the information referred to in point (e), if necessary amend control measures in line with the requirements of Section 4.			 Plans for studies and other activitie about the safety and efficacy of the Amasuring the effectiveness of risk If agreed with the competent authority a management planning purposes, the sainclude additional elements if they are ridentified risk, important potential risk or as: The disposal of the product where because of remaining active substate Innovative pharmaceutical forms (epercentage of active substance wh for patient and related side effects; gastricresident dosage forms for ul may improve patients adherence to the gastro-intestinal side effect); Use of a medical device and risk medical device; Quality aspects relevant in relation and not adequately addressed Key messages: The Risk Management comprises GMP and Post-Market Surve MDR directly refers to Plan and report are a Risk Management is linked to the Post and is mentioned in EN ISO 14971. 	e medicine; -minimisation measures. and where needed for risk afety specification may esulting in important r missing information such it might pose a particular risk ance (e.g. patches); e.g. to contain a higher ich reduces the dose burden long-term delivery tra-long-acting drug delivery tra-long-acting drug delivery to the safety of the product t for Medical Devices eillance (Vigilance). structure. Topics to be in the described.



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Requirem	nents	
				EN ISO 14971 "Medical devices — App to medical devices (ISO 14971:2007, Co 01) is mentioned in The Publication of ti harmonised standards under Union harm US-FDA ISO 14971 is also a recognized consens 21CfR 820 does not directly mention Ris point out Management responsibility to o Quality Management System. Risk Anal validation.	orrected version 2007-10- tles and references of monisation legislation sus standard for the FDA sk Management but does check the adequacy of the
	al Safety and Performance Requirer		1		
Chapter I	In eliminating or reducing risks related to use error, the	NA (Annex I)	EU : EMA/606103/2014	Link with PQS:	
General requirements Section 5	manufacturer shall: (a) reduce as far as possible the risks related to the ergonomic features of the device and the environment in which the device is		US-FDA :21CFR820-30 Design Control	EMA has issued a guideline document of prevention of medication errors: EMA/60 "Pharmacovigilance key principles of ris relation to medication errors arising from	06103/2014 k management planning in
Risks related to use error	intended to be used (design for patient safety). (b) give consideration to the technical knowledge, experience, education, training and use environment, where applicable, and the medical and physical conditions of intended users (design for lay, professional, disabled or other users).			its delivery system". This good practice guide outlines the ker management planning in relation to mee the medicinal product and its delivery sy Please note that medication errors cove <u>US-FDA</u> :21CFR820 30 Design Control	dication errors arising from ystem.



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	PART I – Obligations of the Manufa	cturer and General S	afety and Performance Require	ments	
				The need for human factors is implied: c) Design input – includes "needs of the f) Design validation – performance criting) Design validation – " devices confort and intended uses and shall include test under actual or simulated use conditions include software validation and risk anall FDA Draft Guidance (2016): Human factor clinical study considerations in combinated development. Draft Guidance for Industion Key messages: Question for Single Integral device component is how these checks are documented and what is the basis for the notified bodies to get their opinion on the Drug-Device Combination.	eria met rm to defined user needs ting of production units s. Design validation shall lysis" tors studies and related tion product design and



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
Annex I – Gener	PART I – Obligations of the manufa al Safety and Performance Require	ments – Chapter III		nents	
Requirements Regarding The Information Supplied with the Device (Sections 23.1., 23.2, 23.3 and 23.4)	Each device shall be accompanied by the information needed to identify the device and its manufacturer, and by any safety and performance information relevant to the user, or any other person, as appropriate. Such information may appear on the device itself, on the packaging or in the instructions for use, and shall, if the manufacturer has a website, be made available and kept up to date on the website, taking into account the following: (a) The medium, format, content, legibility, and location of the label and instructions for use shall be appropriate to the particular device, its intended purpose and the technical knowledge, experience, education or training of the intended user(s). In particular, instructions for use shall be written in terms readily	NA (Annex I)	EU Directive 2001/83/EC	security plan (Optional, it device)	arning/precaution of use able and hardware IT network and depends on the type of urer (Not relevant for SI DDC, holders:



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCESS	S PART I – Obligations of the manufa	cturer and general s	afety and performance requiren	nents	
Annex I – Gene	ral Safety and Performance Require	ments – Chapter III			
	understood by the intended user				
	and, where appropriate,				
	supplemented with drawings and				
	diagrams.				
	(b) The information required on				
	the label shall be provided on the				
	device itself. If this is not				
	practicable or appropriate, some				
	or all of the information may				
	appear on the packaging for each				
	unit, and/or on the packaging of				
	multiple devices.				
	(c) Labels shall be provided in a				
	human-readable format and may				
	be supplemented by machine-				
	readable information, such as				
	radio-frequency identification				
	('RFID') or bar codes.				
	(d) Instructions for use shall be				
	provided together with devices. By				
	way of exception, instructions for				
	use shall not be required for class				
	I and class IIa devices if such				
	devices can be used safely				
	without any such instructions and				
	unless otherwise provided for				
	elsewhere in this Section.				



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR PROCES	S PART I – Obligations of the manufa	acturer and general s	safety and performance requiren	ients	
Annex I – Gene	eral Safety and Performance Require	ments – Chapter III			
	(e) Where multiple devices are				
	supplied to a single user and/or				
	location, a single copy of the				
	instructions for use may be				
	provided if so agreed by the				
	purchaser who in any case may				
	request further copies to be				
	provided free of charge.				
	(f) Instructions for use may be				
	provided to the user in non-paper				
	format (e.g. electronic) to the				
	extent, and only under the				
	conditions, set out in Regulation				
	(EU) No 207/2012 or in any				
	subsequent implementing rules				
	adopted pursuant to this				
	Regulation				
	(g) Residual risks which are				
	required to be communicated to				
	the user and/or other person shall				
	be included as limitations, contra-				
	indications, precautions or				
	warnings in the information				
	supplied by the manufacturer				
	(h) Where appropriate, the				
	information supplied by the				
	manufacturer shall take the form				
	of internationally recognised				



MDR Article reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR Part 3 or 4 reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non- Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
	PART I – Obligations of the manufa al Safety and Performance Require		afety and performance requireme	ents	
	symbols. Any symbol or				
	identification colour used shall				
	conform to the harmonised				
	standards or CS. In areas for				
	which no harmonised standards				
	or CS exist, the symbols and				
	colours shall be described in the				
	documentation supplied with the				
	device.				



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process	Part II - Clinical Evaluation				
Article 2 Definition (44) Clinical Evaluation	'clinical evaluation' means a systematic and planned process to continuously generate, collect, analyse and assess the clinical data pertaining to a device in order to verify the safety and performance, including clinical benefits, of the device when used as intended by the manufacturer;	NA	ICH E Guidelines Regulation (EU) No 536/2014	 Only Clinical Study and Clinical Transmission The use of the term Clinical Evaluation clearly defined and covers specifical literature, which is not considered 536/2014. Key message: Clinical Evaluation does not necest (Study). MDR makes clear distinct evaluation of all available clinical is demonstration to GSPR (Annex I) appropriate clinical evaluation. It is recognized that the meeting by NB WG on July 5th 2019 also provided that the meeting by NB W	term designating the assessment tate the safety of Medicinal t used in Regulation EU 536/2014. rials are used. ation under MDR 2017/745 is c meanings like relevant scientific per se in ICH E guidelines or EU ssarily mean Clinical Investigation tion between both terms. A critical nvestigation data and might form the basis of an metween EFPIA/EBE and TEAM-
				as part of the NB assessment to r See also Article 61 Section 3	neet Article 117.



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process I	Part II - Clinical Evaluation				
Article 2	'clinical data' means information	NA	ICH E Guidelines	Link with PQS: The term Clinical Data under MDI	R 207/745 covers more meanings
Definition	concerning safety or performance that is generated		Regulation (EU) No	than the same term under EU 536 clinical data derived from clinical	6/2014. For medicinal products,
(48) Clinical data	from the use of a device and is sourced from the following: — clinical investigation(s) of the device concerned, — clinical investigation(s) or other studies reported in scientific literature, of a device for which equivalence to the device in question can be demonstrated, — reports published in peer reviewed scientific literature on other clinical experience of either the device in question or a device for which equivalence to the device in question can be demonstrated, — clinically relevant information		536/2014		



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process	Part II - Clinical Evaluation				
Article 2 Definition (51) Clinical evidence	Clinical evidence: means clinical data and clinical evaluation results pertaining to a device of a sufficient amount and quality to allow a qualified assessment of whether the device is safe and achieves the intended clinical benefit(s), when used as intended by the manufacturer	NA	Not mentioned in EU 536/2014, nor in US-FDA 21 CFR Part 4 or 820	Key message: This definition uses the term Clini MDR . See definition (53) here-be	cal Benefit which is also defined in elow.
Article 2 Definition (53) Clinical benefit	Clinical benefit: means the positive impact of a device on the health of an individual, expressed in terms of a meaningful, measurable, patient- relevant clinical outcome(s), including outcome(s) related to diagnosis, or a positive impact on patient management or public health	NA	ISO 14971	<u>Key message:</u> This term is mentioned in risk ma ISO 14971.	anagement for Medical Device EN
Article 61 Section 1 Clinical Evaluation	Confirmation of conformity with relevant general safety and performance requirements set out in Annex I under the normal conditions of the intended use of the device, and the evaluation of the undesirable side-effects and of the	Annexe I (GSPR) Annexe III (CONFORMITY TO THE TECHNICAL DOCUMENTATION REQUIREMENTS ON POST- MARKET SURVEILLANCE	NA	Key message: This section points out the require evidence of fulfilling Annex 1 with reference to Annex III and Annex It is important to consider Annex I devices are intended to admini- shall be designed and manufact compatible with the medicinal pro-	a clinical evaluation with XIV (Part A). , Chapter II (10.3): <i>"If the</i> <i>ster medicinal products they</i> <i>stured in such a way as to be</i>



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process	Part II - Clinical Evaluation				
	acceptability of the benefit-risk- ratio referred to in Sections 1 and 8 of Annex I, shall be based on clinical data providing sufficient clinical evidence, including where applicable relevant data as referred to in Annex III. The manufacturer shall specify and justify the level of clinical evidence necessary to demonstrate conformity with the relevant general safety and performance requirements. That level of clinical evidence shall be appropriate in view of the characteristics of the device and its intended purpose. To that end, manufacturers shall plan, conduct and document a clinical evaluation in accordance with this Article and Part A of Annex XIV.	OF THE MEDICAL DEVICES REGULATION 2017/745) Annexe XIV (EVALUATION AND POST- MARKET CLINICAL FOLLOW-UP) Part A (CLINICAL EVALUATION)		with the provisions and restriction. products and that the performance and of the devices is maintained in indications and intended use. This overall requirement implies to not necessarily to generate clinical studies). See also: - Article 5, sections 2 & 3 - Plactinto service. - Article 61 Section 3 and EFPIA Industry. Point of advocacy It is EFPIA advocacy point that the Article 61 requirement for DDC proclinical data delivered with the device. Request for clarifications from Stakeholders: - Annex I does not mention Clinical Evaluation nor Article 61. Clarification is required as DDC clinical data do not necessarily include the use of the single integral device. - What level of evidence would be required by NB, as a function of the device component risk class and on the available clinical evaluation ?	e of both the medicinal products in accordance with their respective o evaluate the clinical aspects but il investigation data (Clinical ing on the market and putting a recommendation to Pharma



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process	Part II - Clinical Evaluation				
Article 61 Section 3 Clinical Evaluation	A clinical evaluation shall follow a defined and methodologically sound procedure based on the following: (a) a critical evaluation of the relevant scientific literature currently available relating to the safety, performance, design characteristics and intended purpose of the device, where the following conditions are satisfied: — it is demonstrated that the device subject to clinical evaluation for the intended purpose is equivalent to the device to which the data relate, in accordance with Section 3 of Annex XIV, and — the data adequately	Annex XIV (EVALUATION AND POST- MARKET CLINICAL FOLLOW-UP) Annex XV (CLINICAL INVESTIGATIONS)	EU Eudralex Volume 1 US-FDA: NA	 and not for medical devices. However, it is mentioned in ISO 1 development validation. <u>Key messages:</u> Clinical Evaluation does not nece (Study). A critical evaluation of investigation data and demons form the basis of an appropriat For Non Integral device, it is key fourpose declared in the Conformitient Request for clarification from S EFPIA would like clarity about diff and novel devices. For novel device should contain Drug Clinical Data device in order to support clinical Post-workshop note: EMA dGC 	ssarily mean Clinical Investigation all available clinical tration to GSPR (Annex I) might e clinical evaluation. to demonstrate that the intended ity Assessment is respected. Stakeholders ferences between well-established vice, the Clinical Evaluation Report , Scientific Data & Literature on the
	demonstrate compliance with the			Recommendation to Pharma In	dustry:



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process	Part II - Clinical Evaluation				
	relevant general safety and performance requirements; (b) a critical evaluation of the results of all available clinical investigations, taking duly into consideration whether the investigations were performed under Articles 62 to 80, any acts adopted pursuant to Article 81, and Annex XV; and (c) a consideration of currently available alternative treatment options for that purpose, if any.			Clinical aspects should be be eml product (Cfr ISO 13485:2016) <u>US-FDA:</u> Clinical Evaluation is no 820.	
Article 61 Section 11 Clinical evaluation Update	The clinical evaluation and its documentation shall be updated throughout the life cycle of the device concerned with clinical data obtained from the implementation of the manufacturer's PMCF plan in accordance with Part B of Annex XIV and the post-market surveillance plan referred to in Article 84. For class III devices and implantable devices, the PMCF evaluation report and, if indicated, the summary of safety and	Annex XIV – Part B (EVALUATION AND POST- MARKET CLINICAL FOLLOW-UP)	NA	Point of advocacy: It is assumed that drug clinical data cover the device constituent for safety and performance, and therefore Art 61 would not apply.	Request for clarification from Stakeholders: Non Integral DDC product would be regulated under medicinal product Directive 2001/83/EC and would therefore be exempt from post-market surveillance plan as referred under MDR in Article 83 & 84. See MDR Process Part V – POST- MARKET SURVEILLANCE, VIGILANCE AND MARKET SURVEILLANCE



MDR Article Reference	MDR Text	MDR Annex reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device constituent)
MDR Process I	Part II - Clinical Evaluation				
	clinical performance referred to in Article 32 shall be updated at least annually with such data.				
ANNEX XV CLIN	ICAL INVESTIGATIONS			·	•
Annex XV Chapter II section 1.7 page 168	if the application is submitted in parallel with an application for a clinical trial in accordance with Regulation (EU) No 536/2014, reference to the official registration number of the clinical trial;	NA (Annex XV)	Regulation (EU) No 536/2014	Link with PQS: This paragraph of Annexe XV, Ch assessment for device used in co based on clinical data generated for clinical trial.	



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
	S PART III - CLASSIFICATION AND C	CONFORMITY ASSES	SSMENT		
Article 51	ON OF DEVICES	Annex VIII	NA		
Alticle 51	Devices shall be divided into	(Classification		Link with PQS: There is no device	e classification system described
Section 1	classes I, IIa, IIb and III, taking into account the intended purpose	rules)		in medicinal product directives.	
Classification	of the devices and			Key messages:	
of devices	their inherent risks. Classification			- Application of the MDR classific	ation rules shall be governed by
	shall be carried out in accordance			the intended purpose of the CE m	arked devices;
General rule	with Annex VIII.			- Single-Integral DDCs are not Me not be classified as per MDR Ann	
				However, EMA has mentioned the rules to the device component of document published on 27 Februa Questions & Answers on Impleme and In Vitro Diagnostic Medical D 2017/745 and (EU) 2017/746));	a Single Integral DDC (See EMA ary 2019, page 5, question 1.3 - entation of the Medical Devices evices Regulations ((EU)
				Therefore, when applying these	•
				Single-Integral DDC, Pharma Indu	-
				component as it would be used a Single-Integral DDC would need o upcoming guidance documents b	confirmation with the issuance of
				Recomendation to Pharma Indu	istry:
				EFPIA recommends Pharma Indu implementing and classification ru marked device classification.	· · · · · · · · · · · · · · · · · · ·
				For ease of review by Pharma Ind series of classification example for marketed DDC products.	



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCES	S PART III - CLASSIFICATIO	ON AND CONFORMITY ASS	ESSMENT		
				Examples for Single-Integral Device component: 1° Pre-filled syringe – Class IIa (Rule 6) (**) if the administration of the medicinal product is done in a manner that is not potentially hazardous; Otherwise Class IIb. 2° Patch – Class I (Rule 2) (*)	Examples for CE marked Device: 1° Vial adaptor – Class IIa (Rule 2) (*) 2° Reconstitution syringe – Class IIa (Rule 2) (*) 3° Stand-alone syringes with needles - Class IIa (Rule 6) (**) 4° Electronic injector – Class IIa (Rule 12) (***) 5° Nazal or buccal syringe for oral administration – Class IIa (Rule 20) (****)
				 storing blood, body liquids, cell for the purpose of eventual infuintroduction into the body are of a control of the purpose of eventual infuintroduction into the body are of a control of the purpose of eventual infuintroduction into the body are of a control of the purpose of	usion, administration or classified as class IIa: class IIa, class IIb or class III channeling or storing blood or rgans, parts of organs or body d bags; blood bags are classified ch devices are classified as class ive devices intended for class IIa unless they are intended



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III - CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
				is potentially hazardous taking ac in which case they are classified a	
				in a manner that is potentially haz nature of the substances involved concerned and of the mode of ap classified as class IIb. (****) Rule 20: Invasive device f product by inhalation - All invasi	dy liquids or other substances ed as class IIa, unless this is done cardous, taking account of the d, of the part of the body plication in which case they are for administration of medicinal sive devices with respect to cally invasive devices, which are al products by inhalation are eir mode of action has an nd safety of the administered nded to treat life- threatening
CONFORMITY A	SSESSMENT				
Article 52	Prior to placing a device on the market , manufacturers shall	Annex II (Technical Documentation)	NA	Key messages :	
Sections 1 & 2 Conformity assessment	undertake an assessment of the conformity of that device, in accordance with the applicable conformity assessment	Annex III (Technical Documentation post-market		 There is no need for Pharma Ir comply with conformity assessme Integral and Co-Packaged DDCs product directives, and CE marke are supplied by Device Manufacture 	ent procedures, since both Single- are governed by medicinal d devices for Co-Packaged DDCs
procedures	procedures set out in Annexes IX to XI.	Annex IX (Conformity		However, there is a need for Phar Medical Device QMS aspects as document, in order to:	rma Industry to understand
	Prior to putting into service a device that is not placed on the	Assessment Based on QMS and Assessment of		 Use a streamline approach w similar to US-FDA 21 CFR P Be able to define mutual exp 	when developing DDC products, art 4 , ectations and responsibilities in act between Pharma Company



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III - CLASSIFICATION AND C		SMENT		
	market, manufacturers shall undertake an assessment of the conformity of that device, in accordance with the applicable conformity assessment procedures set out in Annexes IX to XI.	Technical Documentation) Annex X (Examination of assessment) Annex XI (QMS for Product Conformity Verification)		Manufacturer). The Authors underline that the QM Article 52 are developed as a fund The interesting information for Ph Annex IX and XI which describe in for Device Manufacturers and sur function of Device classification.	ction of the Device class. arma Industry is the reference to n more details QMS requirements veillance by NBs (Audits) as a preport PMS & vigilance to Device nsibilities and process for this Quality Agreement. There is a fety Data exchange between
Article 55 Clinical evaluation consultation procedure for certain class III and class IIb devices	In addition to the procedures applicable pursuant to Article 52, a notified body shall also follow the procedure regarding clinical evaluation consultation as specified in Section 5.1 of Annex IX or as referred to in Section 6 of	Annex VIII (Classification rules) Annex IX (Conformity Assessment Based on QMS and Assessment of	NA	Key messages : 1) Article 55 applies to Class III in delivery devices. Note: Class III w which the drug has the primary m 2) MDR regulates expert panels a 106 and 107.	ould not be relevant to DDC for ode of action.



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
Section 1	Annex X, as applicable, when performing a conformity assessment of the following devices: (a) class III implantable devices, and (b) class IIb active devices intended to administer and/or remove a medicinal product, as referred to in Section 6.4 of Annex VIII (Rule 12).	Technical Documentation) Annex X (Examination of assessment)		Key message :The use of classification rules for device, mentioned by EMA in its Q & A document (28 Feb 2019) does not mean that Single-Integral DDC, falling into class IIB for its device component, would be reviewed by the Expert Panel regarding clinical evaluation consultation. Indeed, EMA states that specific impact of the device on the medicinal product is under the authority of CA/EMA.	Key message : Class IIb device used in Non integral DDC might impose the consultation of the Expert Panel by NB. Recommendation to Pharma Industry: It is recommended to identify and engage in discussions with a NB and to engage with medicines CAs in a timely manner, in order to confirm the device classification and the procedure for clinical evaluation.
Article 56 Declaration of Conformity Sections 1 & 2	The certificates issued by the notified bodies in accordance with Annexes IX, X and XI shall be in an official Union language determined by the Member State in which the notified body is established or otherwise in an official Union language acceptable to the notified body. The minimum content of the certificates shall be as set out in Annex XII.	Annex IX (Conformity Assessment Based on QMS and Assessment of Technical Documentation) Annex X (Examination of assessment) Annex XI (QMS for Product Conformity Verification)	NA	Point of advocacy:	ns and Conditions how long his no consistency ensured among and max 5 years for CE marking. Procedure, when considering no or ance or rule, using a risk-based minimum validity of a certificate.



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
	The certificates shall be valid for the period they indicate, which shall not exceed five years. On application by the manufacturer, the validity of the certificate may be extended for further periods, each not exceeding five years, based on a re-assessment in accordance with the applicable conformity assessment procedures.	Annex XII (Certificates issued by a Notified Body)			
Article 86 Periodic safety update report (3)	For devices other than those referred to in paragraph 2 (i.e. Class III devices), manufacturers shall make PSURs available to the notified body involved in the conformity assessment and, upon request, to competent authorities.	NA	Medicinal Product Directive 2001/83/EC	Link with PQS: Article 86 is not applicable to DDCs registered as medicinal products. Request for clarification However, it is not clear yet to what extent Pharma Company would need to provide data PSUR to Device Manufacturer for the Device used in Non Integral DDCs. EFPIA requires clarification from	
Article 117 Amendment to Directive 2001/83/EC '(12)	In Annex I to Directive 2001/83/EC, point 12 of Section 3.2. is replaced by the following: '(12) Where, in accordance with the second subparagraph of Article 1(8) or the second subparagraph of Article 1(9) of Regulation (EU) 2017/745 of the European Parliament and of the Council (*), a product is governed by this Directive, the marketing authorisation dossier shall include,	Annex I (GSPR)	Medicinal Product Directive 2001/83/EC	Stakeholders. Link with PQS: MDR Article 117 includes Integral combination products in its scope included in Article 117, as per MD products described in Article 1 Series For Single-Integral DDCs, the cor partially addressed by EMA in its Integral DDCs.	Non Integral DDCs are not R scope and matter for DDC ctions 8 & 9.



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCES	S PART III - CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
	where available, the results of the assessment of the conformity of the device part with the relevant general safety and performance requirements set out in Annex I to that Regulation contained in the manufacturer's EU declaration of conformity or the relevant certificate issued by a notified body allowing the manufacturer to affix a CE marking to the medical device.			- Timing for establishing N	at and the process by which NB Integral DDC. that: Intent of NBOp, not on QMS; IBOp would be NB dependent; Ind CA/EMA respective GSPR data to f data might be necessary for
	If the dossier does not include the results of the conformity assessment referred to in the first subparagraph and where for the conformity assessment of the device, if used separately, the involvement of a notified body is required in accordance with			Request for clarification from S case: For the submission of a new Sing May 2020, would a relevant certifi allowing the manufacturer to affix device under the MDD be accepta the certificate is still valid ?	le Integral DDC product after 26 cate issued by a notified body
	Regulation (EU) 2017/745, the authority shall require the applicant to provide an opinion on the conformity of the device part with the relevant general safety and performance requirements set out			Point of advocacy: EFPIA would like clear process at made public, how fees are derived and Pharma Industry can use the engagement process (Cfr Art. 53,	d, and how Device Manufacturer se information in the NB
	in Annex I to that Regulation issued by a notified body designated in accordance with that Regulation for the type of device in question.			Recommendation to Pharma In It is recommended to carefully rev applicant-NB contract when engage	iew and define timelines in the



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
	(*) Regulation (EU) 2017/745 of the European Parliament and of the Council of 5 April 2017 on medical devices, amending Directive 2001/83/EC, Regulation (EC) No 178/2002 and Regulation (EC) No 1223/2009 and repealing Council Directives 90/385/EEC and 93/42/EEC (OJ L 117, 5.5.2017, p. 1).				
ANNEX IX CON	FORMITY ASSESSMENT BASED ON	QMS AND ASSESS	MENT OF TECHNICAL DOCUMEN	NTATION	
Chapter I	(c) the procedures and techniques for monitoring, verifying, validating	NA	NA	Key messages:	
Section 2.2 (c)	and controlling the design of the devices and the corresponding				on integral DDCs, EMA states in uirements for DDCs, under section es 169-171), that <i>Compliance of a</i>
Design of the device	documentation as well as the data and records arising from those procedures and techniques. Those procedures and techniques shall specifically cover:			DDC with relevant Ph. Eur. cha be demonstrated. Ph.Eur. require guidance take precedence over IS	pter(s) or monograph(s) should ements and European and ICH
	— the strategy for regulatory compliance, including processes			2) MDR 2017/745 places Design for Medical Device.	of the device at the core of QMS
	for identification of relevant legal requirements, qualification,			Of particular interest are:	
	classification, handling of equivalence, choice of and compliance with conformity assessment procedures,			 a) The need to define the regulate b) The need to identify the application these requirements (Applicable C adequate solutions) For instance, ISO10993 	able GSPR and solutions to fulfill S, harmonized standards or other



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	S PART III - CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
	 identification of applicable general safety and performance requirements and solutions to fulfil those requirements, taking applicable CS and, where opted for, harmonised standards or other adequate solutions into account, risk management as referred to in Section 3 of Annex I, the clinical evaluation, pursuant to Article 61 and Annex XIV, including post-market clinical follow-up, solutions for fulfilling the applicable specific requirements regarding design and construction, including appropriate pre-clinical evaluation, in particular the requirements of Chapter II of Annex I, solutions for fulfilling the applicable specific requirements regarding the information to be supplied with the device, in particular the requirements of Chapter II of Annex I, solutions for fulfilling the applicable specific requirements regarding the information to be supplied with the device, in particular the requirements of Chapter III of Annex I, the device identification procedures drawn up and kept up to date from drawings, 			 of devices) Risk of patient not administra Manufacturing risk analysis Risk from potential contamina design, manufacturing, packa patient d) The clinical evaluation, pursuar including post-market clinical follo e) The need for pre-clinical evalua (GSPR) Recommendation to Pharma In Using Design Control as per ISO Part 820, or as per MDR Annex I development of a new Single Inte including Usability Engineering, p path, to facilitate compliance with 	arrors (e.g. due to some similarity ting the correct dose ation arising from the device aging storage and shipment to the ation as per Chapter II of Annex I ation as per Chapter II of Annex I dustry: 13485:2016, or as per 21 CFR X, right from the beginning of the gral Product or Non-Integral DDC, rovides a structured development regulatory requirements as ance on Quality Requirements for



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
Chapter I Section 4.10 Change of an approved device	specifications or other relevant documents at every stage of manufacture, and — management of design or quality management system changes; Changes to the approved device shall require approval from the notified body which issued the EU technical documentation assessment certificate where such changes could affect the safety and performance of the device or the conditions prescribed for use of the device. Where the manufacturer plans to introduce any of the above- mentioned changes it shall inform the notified body which issued the EU technical documentation assessment certificate thereof. The notified body shall assess the planned changes and decide whether the planned changes require a new conformity assessment in accordance with Article 52 or whether they could	NA	EUDRALEX Vol 4 EU 2001/83/EC Medicinal Product Directives EMA draft Guidance on Quality Requirements for DDC (EMAdGQR-DDC)	of the appropriate category to CA(s). All changes to me components within DDCs shou with the relevant EU Variation variation guidelines in place and appropriate category. Depending on the nature of consider whether updates to NBOp, Declaration of Conform with the device in question are The category of variation shou impact of the change, e.g. a cha DDC CQAs and/or any element strategy may be considered a cases where the need for a	xpectations in Section 8 – guideline will require a variation be submitted to the medicines edical devices and/or device and be presented in accordance ons Regulation and associated d should be submitted under the the change, the MAH should or relevant documentation (e.g. mity, CE mark etc.) associated e required to support the change. uild take into consideration the ange to a device that impacts any mt(s) of the overall DDC control higher category of variation. In variation is unclear and/or the lear, it is recommended that the
	be addressed by means of a supplement to the EU technical documentation assessment certificate. In the latter case, the				e MA is consulted to agree the



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	CONFORMITY ASSES	SMENT		
	notified body shall assess the changes, notify the manufacturer of its decision and, where the changes are approved, provide it with a supplement to the EU technical documentation assessment certificate.			Key message : The medicines CA are the main both Single-Integral and Non Inte Request for clarifications from EFPIA would like clarifications wit DDC change types in EMA variati II).	gral DDCs. <mark>Regulators:</mark> h regards to the translation of
ANNEX XIV CLI	NICAL EVALUATION AND POST-M	ARKET FOLLOW-UP			
PART B POST-MARKET CLINICAL FOLLOW-UP (5)	PMCF shall be understood to be a continuous process that updates the clinical evaluation referred to in Article 61 and Part A of this Annex and shall be addressed in the manufacturer's post-market surveillance plan. When conducting PMCF, the manufacturer shall proactively collect and evaluate clinical data from the use in or on humans of a device which bears the CE marking and is placed on the market or put into service within its intended purpose as referred to in the	NA	EUDRALEX VOL 4	Link with PQS: The Device Manufacturer of a No considered as a supplier of the PI Although it is not clear yet to wha to feed in the PMCF plan of the D agreement (QAA) shall describe t Pharma Company and Device Ma pertinent data for updates of clinic data should be defined for both di	narma Company. t extent Pharma Industry will need evice Manufacturer, the quality he mutual responsibilities of anufacturer with regards to cal evaluations. The exchange of



MDR Article Reference	MDR Text	MDR Annex Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART III – CLASSIFICATION AND C	ONFORMITY ASSES	SMENT		
	relevant conformity assessment				
	procedure, with the aim of				
	confirming the safety and				
	performance throughout the				
	expected lifetime of the device, of				
	ensuring the continued				
	acceptability of identified risks and				
	of detecting emerging risks on the				
	basis of factual evidence.				



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
The Guideline EN	PART IV – REGISTRATION PROCES	provides guidance on th	ne documentation expected for Dru	ug-Device Combinations (DDCs) in t	he quality part of the dossier for a
Article 29 Section 1 Registration of devices	than a custom-made device, on the market, the manufacturer shall, in accordance with the rules of the issuing entity referred to in	than a custom-made device, on the market, the manufacturer shall, in accordance with the rules(European UDI system)US-FDA 21 CFR Part 820.120Part B		labeling needs to follow med	oduct does not require UDI; DDC licinal product regulations and er, CE marked devices have UDI
	DI as defined in Part C of Annex VI to the device and shall provide it to the UDI database together with the other core data elements referred to in Part B of Annex VI related to that device.	ELEMENTS TO BE PROVIDED TO THE UDI DATABASE TOGETHER WITH THE UDI-DI IN ACCORDANCE WITH ARTICLES 28 AND 29) Part C (THE UDI SYSTEM)		Key message: Single-Integral DDCs are not subjected to MDR requirements for UDI. This was confirmed by MDCG (Guidance 2019-2, published in February 2019).	Key message: CE marked devices, used in Non Integral DDCs, have UDI affixed, as per MDR Article 29. Point for advocacy The PQS ensures that the device information is traceable; EFPIA advocates for UDI being affixed on the device itself or its primary packaging, not on the secondary packaging of the DDC, similar to US-FDA 21 CFR Part820 (See below). The UDI will be checked and documented in DDC batch records and SmPC. Non Integral DDCs registered as medicinal porduct must indeed comply



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device					
	IDR PROCESS PART IV - REGISTRATION PROCESS									
					with serialisation for safety measure and traceability along the distribution chain.					
					US-FDA 21cfr820.120 requires an UDI for medical Devices: However there is an exemption (b) National Drug Code (NDC) Numbers. If a combination product properly bears an NDC number on its label (1) The combination product is not subject to the requirements of 801.20. (2) A device constituent of such a combination product whose components are physically, chemically, or otherwise combined or mixed and produced as a single entity as described by 3.2(e)(1) of this chapter is not subject to the requirements of 801.20. (3) Each device constituent of such a combination product, other than one described by 3.2(e) (1) of this chapter, must bear a UDI on its label unless paragraph (a) (11) of this					



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS F	PART IV - REGISTRATION PROCES	SS			
Article 30 Section 2	Member States may maintain or introduce national provisions on registration of distributors of devices which have been made available on their territory.	NA	NA	Key message: This requirement is not applicable to Pharma Industry marketing and distributing Single-Integral DDC.	Point for advocacy On top of EFPIA request for streamline approach of GDP for Non Integral DDC, i.e., with limited application of Article 29, EFPIA advocates for non- applicability of Articles 14 & 30 to Pharma Company distributing Non Integral DDCs.
Article 31 Sections 1 & 2 Registration of manufacturers, authorised representatives and importers	Before placing a device, other than a custom-made device, on the market, manufacturers, authorised representatives and importers shall, in order to register, submit to the electronic system referred to in Article 30 the information referred to in Section 1 of Part A of Annex VI, provided that they have not already registered in accordance with this Article. In cases where the conformity assessment procedure requires the involvement of a notified body pursuant to Article 52, the information referred to in Section 1 of Part A of Annex VI shall be provided to that electronic system before applying to the notified body.	Annex VI (European UDI system) Part A INFORMATION TO BE SUBMITTED UPON THE REGISTRATION OF DEVICES AND ECONOMIC OPERATORS IN ACCORDANCE WITH ARTICLES 29(4) AND 31	NA	Key message: These requirements are only pert would import CE marked devices	



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device
MDR PROCESS	PART IV – REGISTRATION PROCE	SS			
	After having verified the data entered pursuant to paragraph 1, the competent authority shall obtain a single registration number ('SRN') from the electronic system referred to in Article 30 and issue it to the manufacturer, the authorised representative or the importer.				



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)
MDR PROCESS	PART V – POST-MARKET SURVEIL	LANCE, VIGILANCE	AND MARKET SURVEILLANCE		
Article 83 Section 1 Post-market surveillance system of the Manufacturer	For each device, manufacturers shall plan, establish, document, implement, maintain and update a post-market surveillance system in a manner that is proportionate to the risk class and appropriate for the type of device. That system shall be an integral part of the manufacturer's quality management system referred to in Article 10(9) (= General Obligation of the Manufacturer – QMS)	NA	Medicinal product directive 2001/83/EC	to the competent authority for mea The Device manufacturer will do t role and responsibilities should be QAA should include reporting resp specified in MDR. PQS has also similar requirement product than the requirements set	 products, Pharma Company nt Authority (CA) only. Therefore 88, 89 & 95 would not apply to and marketing DDCs. a to the supplier of the device, and dicinal product. he post- market surveillance. This e in the quality agreement. The ponsibilities and timing as s for Distributor of medicinal forth in MDR Article 14 Section 5 s for market complaints). See also specific to Non Integral DDC. takeholders: d be clear for DDCs registered as t Surveillance System is required FETY AND PERFORMANCE heral Requirements, Section 3.e. ext need to establish a PMS or DDC products as described in



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)
MDR PROCES	S PART V – POST-MARKET SURVE	LLANCE, VIGILANCE	AND MARKET SURVEILLANCE		
				EFPIA will work with the Stakehol regards to applicability of MDR Ar	0
				 can arise from User not using the device defect (Manufacturing cause device. It is important to define the expectations clearly with the legal Third party QAA. Pharma should continuously eval for understanding of the safety, 	al knowledge and processes in s appropriately. Market complaints device as appropriate, from a se) or from the design of the e market complaint investigation
				Key message (Input from EFPIA-MPP-MFE survey):- For SI DDC, reporting to the competent authority would mean reporting to CAMS / EMA only. There is no provision for double reporting as it is the case in Japan and US where device directly related post-market surveillance is required.Recommendations to Pharma Industry, specific to SI DDC:	Key message (Input from EFPIA-MPP-MFE survey): Adverse event (AE) caused by a Co-packaged DDC would be reported to the respective competent authority: The device part would be reported to both EMA/CA and CAMD, by the MAH to EMA/CA and by the legal manufacturer of the device to CAMD. This requirement implies to define the data flow and confidentiality requirements in quality agreement (QAA)



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)					
MDR PROCES	IDR PROCESS PART V – POST-MARKET SURVEILLANCE, VIGILANCE AND MARKET SURVEILLANCE									
				(As an input from EFPIA-MPP survey) Should the Pharma company perform an annual market surveillance for the device component as per Annex II which refers to article 83-86 ? The response was no, since SI DDC must comply with MDR Annex I only. However, recommendation is to do it and keep it internally if not required by NB. Requirement might depend on the device risk profile.	approved between the Pharma Company and the legal manufacturer of the device. - It is also assumed that Pharma Company would not be considered as Distributor under MDR Article 14. See point for advocacy here-below. Point for advocacy (EFPIA & <u>MFE):</u> It is EFPIA and MFE point for advocacy that MDR Article 14 Section 5 should not apply to Pharma Company marketing and distributing Device that is Co-Packed with a medicinal drug, since the DDC is governed by Directive 2001/83/EC or Regulation (EC) No 726/2004, as applicable. See also MDR Process Part VI –					
					by Directive 2001 Regulation (EC) as applicable. Se					



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)					
MDR PROCESS	MDR PROCESS PART VI – OBLIGATIONS OF THE OTHER ECONOMIC ACTORS (Distributors and Importers)									
Article 13 General obligations of Importers Sections 1 to 10	Importers shall place on the Union market only devices that are in conformity with this Regulation.	NA	Directive 2001/83/EC, Regulation (EC) No 726/2004	Link with PQS: Pharma Company importing Single-Integral DDC must comply with EC Directives and Regulations pertinent to medicinal products, and with Article 1 Section 9, Article 117 and Annex I (GSPR). Therefore Article 13 does not apply to SI DDC.	Key message: Pharma Company importing CE marked Device into EU must comply with MDR Article 13, Article 25 (Identification within the supply chain), Article 27 (Storage of UDI data), Article 30 Para 3 (Verify registration of Manufacturer, Authorised Representative; Add in Eudamed own details), Article 31 (Registration of Importer SRN). All these requirements are not mapped against medicinal product directives since CE marked Device must comply with new MDR 2017/745 requirements. It is therefore Pharma Company considered as an Importer to ensure compliance with MDR requirements. Pharma Company, considered as an Importer under MDR, would then need to nominate or establish contract agreement with an Authorized Representative. The latest represents the Device legal manufacturer located outside of EU and must comply with					



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)					
MDR PROCESS	IDR PROCESS PART VI – OBLIGATIONS OF THE OTHER ECONOMIC ACTORS (Distributors and Importers)									
Article 14 General obligations of Distributors Section 1 (Comments also valid for Article 14 Sections 3,, 4, 5 & 6)	In the context of their activities, when making a device available on the market, distributors shall act with due care in relation to the requirements applicable.	NA	Directive 2001/83/EC, Regulation (EC) No 726/2004, Commission Delegated Regulation 2016/161 For Safety Features	Link with PQS: Pharma Company distributing Single-Integral DDC must comply with EC Directives and Regulations pertinent to medicinal products. Therefore Article 14 does not apply to SI DDC.	several MDR articles, especially Article 11 which defines the minimum requirements for Authorised Representative. Request for clarification from Stakeholders: EFPIA will work with EMA and Medical Device Regulators to clarify the applicability of MDR for Pharma Company importing Non Integral, co-packaged, DDC. Request for clarifications from Regulators: Applicability of MDR Articles, such as Article 14, beyond CE marking would require clarification from Regulators. It is EFPIA opinion that there is no need for Pharma Company distributing Non Integral Co- Packaged DDC to be considered as Distributor under MDR Article 14, since the CE marked Device is co-packaged with the medicinal product and must be distributed as per GDPs, including as per Regulation 2016/161 for Safety Features. Therefore Complying with requirements set forth in Article 14 is ensured by default.					



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)						
MDR PROCESS	IDR PROCESS PART VI – OBLIGATIONS OF THE OTHER ECONOMIC ACTORS (Distributors and Importers)										
					This is valid for all Sections of Article 14 (1, 3, 4, 5 & 6), with a specific point for advocacy related to UDI check (See Article 14 Section 2 here-below).						
Article 14 General obligations of Distributors Section 2	Before making a device available on the market, distributors shall verify that all of the following requirements are met: (a) the device has been CE marked and that the EU declaration of conformity of the device has been drawn up; (b) the device is accompanied by the information to be supplied by the manufacturer in accordance with Article 10(11); (c) for imported devices, the importer has complied with the requirements set out in Article 13(3); (d) that, where applicable, a UDI has been assigned by the manufacturer. In order to meet the requirements referred to in points (a), (b) and (d) of the first subparagraph the distributor may apply a sampling method that is representative of	NA	Directive 2001/83/EC, Regulation (EC) No 726/2004, Commission Delegated Regulation 2016/161 For Safety Features	NA	Points for Advocacy:Requirements set forth under(a), (b), (c) & (d) are normalrequirements for a PharmaCompany operating under thePQS: Purchasing a CE markedDevice, combining it with amedicinal product and placing iton the market require controls atreception of the Device, duringmanufacturing of the NonIntegral DDC and before QPbatch certification.Therefore EFPIA wouldrecommend an interpretativeguidance, which would allow forDDC product, registered asmedicinal product, to be exemptof the requirements set forth inArticle 14.If Pharma Company would beconsidered as Distributor underMDR Article 14, then EFPIAwould advocate for UDI check						



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)					
MDR PROCESS	IDR PROCESS PART VI – OBLIGATIONS OF THE OTHER ECONOMIC ACTORS (Distributors and Importers)									
	the devices supplied by that distributor. Where a distributor considers or has reason to believe that a device is not in conformity with the requirements of this Regulation, it shall not make the device available on the market until it has been brought into conformity, and shall inform the manufacturer and, where applicable, the manufacturer's authorised representative, and the importer. Where the distributor considers or has reason to believe that the device presents a serious risk or is a falsified device, it shall also inform the competent authority of the Member State in which it is established.				performed at the DDC assembly site <u>only</u> , and not along the supply chain (Wholesalers, hospital,), since it would not be feasible (UDI on primary packaging), or would require UDI on each packaging (First, second, third,).					
Article 16 Section 1 Cases in which obligations of manufacturers apply to	A distributor, importer or other natural or legal person shall assume the obligations incumbent on manufacturers if it does any of the following: (a) makes available on the market a device under its name,	NA	NA	Key message: This MDR Article is not pertinent to Single-Integral DDC.	Key message: Article 1 16 (2b), 16 (3) and 16 (4) should be applied as long as an individual sales unit is separated into individual units (typically this is seen with single-					
importers, distributors or other persons	registered trade name or registered trade mark, except in				use syringes).					



MDR Article Reference	MDR Text	MDR Annex and Reference	Eudralex Vol 4 or ICH or 21 CFR reference	Comments for Single Integral DDC (With device being not CE marked)	Comments for Non-Integral DDC (Where Pharma company is not considered as legal manufacturer of device)
	Cases where a distributor or importer enters into an agreement		ACTORS (Distributors and Impo	rters)	As per alignment obtained between 3 Industry Association
	 importer enters into an agreement with a manufacturer whereby the manufacturer is identified as such on the label and is responsible for meeting the requirements placed on manufacturers in this Regulation; (b) changes the intended purpose of a device already placed on the market or put into service; (c) modifies a device already placed on the market or put into service in such a way that compliance with the applicable requirements may be affected. The first subparagraph shall not apply to any person who, while not considered a manufacturer as defined in point (30) of Article 2, assembles or adapts for an individual patient a device already on the market without changing its intended purpose. 				 (EFPIA-MPP-MFE) in March 2019, it is our understanding that a Pharma company does not become a manufacturer as per MDR Article 16 while copackaging of DDC requires to re-pack the CE marked device, as long as the following criteria are met: There is no change of device intended use, An agreement with the device manufacturer would prevent the application of article 16 as the supplier would meet Pharma needs.